#### UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-Q

#### [X] Quarterly Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

For the quarterly period ended: June 30, 1997

[] Transition Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Commission file number: 1-8443

TELOS CORPORATION (Exact name of registrant as specified in its charter)

Maryland 52-0880974 (State of Incorporation) (I.R.S. Employer Identification No.)

19886 Ashburn Road, Ashburn, Virginia	20147-2358
(Address of principal executive offices)	(Zip Code)

## Registrant's Telephone Number, including area code: (703) 724-3800

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. YES  $_X$  NO \_\_\_\_\_ filing requirements for the past 90 days.

As of August 1, 1997, the registrant had 23,076,753 shares of Class A Common Stock, no par value, and 4,037,628 shares of Class B Common Stock, no par value; and 3,595,586 shares of 12% Cumulative Exchangeable Redeemable Preferred Stock par value \$.01 per share, outstanding.

No public market exists for the registrant's Common Stock.

Number of pages in this report (excluding exhibits): 14

# TELOS CORPORATION AND SUBSIDIARIES

# INDEX

# PART I. FINANCIAL INFORMATION

Item	1.	Financial Statements (Unaudited):
		d Consolidated Statements of Income for the Three and Six Months ed June 30, 1997 and 19963
		d Consolidated Balance Sheets as of June 30, 1997 December 31, 19964
		d Consolidated Statements of Cash Flows for the Months Ended June 30, 1997 and 19965
	Notes to	Condensed Consolidated Financial Statements6-8
Item	2.	Management's Discussion and Analysis of Financial Condition and Results of Operations
		PART II. OTHER INFORMATION

Item	3.	Defaults Upon Senior Securities13
Item	4.	Submission of Matters to a Vote of Security Holders13
Item	6.	Exhibits and Reports on Form 8-K13
SIGNA	TURES	

# PART I - FINANCIAL INFORMATION

## TELOS CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF INCOME (Unaudited) (amounts in thousands)

	Three Mon June		Six Months Ended June 30,	
	1997	1996	1997	1996
Sales Systems and Support Services Systems Integration	\$33,182 24,907	\$26,223 17,643	\$59,300 53,134	\$52,455 31,574
Costs and expenses Cost of sales Selling, general and administrative expenses Goodwill amortization	58,089 48,511 6,869 209	43,866 38,352 6,770 275 	112,434 95,159 13,394 434	84,029 74,255 12,821 550
Operating income (loss)	2,500	(1,531)	3,447	(3,597)
Other income (expenses) Other income (expenses) Interest expense	11 (1,883)	(352) (1,336)	23 (3,643)	(349) (2,536)
Income (loss) before taxes	628	(3,219)	(173)	(6,482)
Income tax provision		-		
Income (loss) from continuing operations	628	(3,219)	(173)	(6,482)
Discontinued operations:				
Income from discontinued operations		261		130
Net income (loss)	\$628 ===	\$(2,958) =====	\$(173) ===	\$(6,352) =====

The accompanying notes are an integral part of these condensed consolidated financial statements.

#### TELOS CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (Unaudited) ASSETS (amounts in thousands)

(amounts	in	thousands)	
----------	----	------------	--

	June 30, 1997	December 31, 1996
Current assets Cash and cash equivalents (includes restricted cash of \$327 at June 30,	\$ 757	\$ 2,781
Accounts receivable, net	49,032	51,549
Inventories, net	15,673	17,066
Other current assets	4,086	2,567
Total current assets	69,548	73,963
Property and equipment, net of accumulated depreciation of		
\$21,580 and \$20,390, respectively	16,536	16,486
Goodwill	13,111	13,545
Other assets	6,553	6,070
	 \$105,748	 ¢110_064
	\$105,748	\$110,064 ======
	LIABILITIES AND STOCKHOLDERS' INVESTMENT	
Current liabilities		
Accounts payable	\$17,544	\$35,730
Other current liabilities	10,281	11,708
Accrued compensation and benefits	8,857	10,163
·	, 	
Total current liabilities	36,682	57,601
Senior credit facility	33,421	15,418
Subordinated notes	16,647	17,439
Capital lease obligation	12,257	12,537
Other long-term liabilities		154
Total liabilities	99,007	103,149
Redeemable preferred stock		
Senior redeemable preferred stock	4,993	4,828
Class B redeemable preferred stock	11,501	11,087
Redeemable preferred stock	27,083	24,230
Total preferred stock	43,577	40,145
Stockholders' investment		
Common stock	78	78
Capital in excess of par	615	4,048
Retained earnings (deficit)	(37,529)	(37,356)
		(0,,000)
Total stockholders' investment (d		(33,230)
	\$105,748	\$110,064
	======	======

The accompanying notes are an integral part of these condensed consolidated financial statements.

# TELOS CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited) (amounts in thousands)

		Months June 30, 1996
Operating activities: Net (loss) income Adjustments to reconcile net (loss) income to cash used in operating activities:	\$ (173)	\$(6,352)
Depreciation and amortization Goodwill amortization Provision for legal settlement Other noncash items Changes in assets and liabilities	1,919 434  39 (19,128)	1,365 780 355 90 (1,024)
Cash used in operating activities	(16,909)	(4,786)
Investing activities: Investment in products Purchase of property and equipment Cash used in investing activities	(1,154) (1,106)  (2,260)	(668) (1,643)  (2,311) 
Financing activities: Proceeds from borrowings under senior credit facility Proceeds from capital lease transaction Payments under capital leases Repayment of senior subordinated notes	18,003 (183) (675)	10,823 1,300 
Cash provided by financing activities	17,145	12,123
(Decrease) increase in cash and cash equivalents Cash and cash equivalents at beginning of period	(2,024) 2,781	5,026 735
Cash and cash equivalents at end of period	\$    757 ====	\$5,761 =====

The accompanying notes are an integral part of these condensed consolidated financial statements.

#### TELOS CORPORATION AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

# Note 1. General

The accompanying condensed consolidated financial statements of Telos Corporation ("Telos") and its wholly owned subsidiaries, Telos Corporation (California), Telos Field Engineering, Inc., Telos International Corporation, and enterWorks.com, inc. (collectively, the "Company") have been prepared without audit. Certain information and note disclosures normally included in the financial statements presented in accordance with generally accepted accounting principles have been condensed or omitted. The Company believes the disclosures made are adequate to make the information presented consistent with past practices. However, these condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Company's annual report on Form 10-K for the fiscal year ended December 31, 1996.

In the opinion of the Company, the accompanying condensed consolidated financial statements reflect all adjustments and reclassifications (which include only normal recurring adjustments) necessary to present fairly the financial position of the Company as of June 30, 1997 and December 31, 1996, and the results of its operations and its cash flows for the three and six month periods ended June 30, 1997 and 1996. Interim results are not necessarily indicative of fiscal year performance because of the impact of seasonal and short-term variations.

In December 1996, the Company sold substantially all of the assets of its consulting division, Telos Consulting Services (TCS), to COMSYS Technical Services, Inc., a subsidiary of COREStaff, Inc. for approximately \$31.6 million. The sale of TCS was treated as a discontinued operation in accordance with APB Opinion Number 30. Accordingly, the results of operations for TCS included in the three and six month period ended June 30, 1996 have been reported separately as "income from discontinued operations".

Certain reclassifications have been made to the prior year's financial statements to conform to the classifications used in the current period.

#### Note 2. Accounts Receivable

The components of accounts receivable are as follows (in thousands):

	June 30, 1997	December 31, 1996
Billed accounts receivable Unbilled accounts receivable	\$35,915 14,164	\$40,225 12,249
Allowance for doubtful accounts	50,079 (1,047)	52,474 (925)
	\$49,032 =====	\$51,549 ======

#### TELOS CORPORATION AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Note 3. Debt Obligations

Senior Credit Facility

At June 30, 1997, the Company had a \$45 million senior credit facility ("Facility") with a bank maturing on July 1, 2000.

#### Note 4. Preferred Stock

#### Senior Redeemable Preferred Stock

The components of the senior redeemable preferred stock are Series A-1 and Series A-2 redeemable preferred stock each with \$.01 par value and 1,250 and 1,750 shares authorized, issued and outstanding, respectively. From July 1, 1995 through June 30, 1997, the Series A-1 and A-2 each carry a cumulative dividend rate equal to 11.125% per annum of its liquidation value, and increases to 14.125% per annum thereafter. The dividends are payable semi-annually on June 30 and December 31 of each year. The liquidation preference of the preferred stock is the face amount of the Series A-1 and A-2 (\$1,000 per share), plus all accrued and unpaid dividends. The Series A-1 and A-2 (\$1,000 per share), plus all accrued and unpaid dividends. The Series A-1 and A-2 Preferred Stock is senior to all other present and future equity of the Company. The Company is required to redeem all of the outstanding shares of the Series A-1 and A-2 on December 31, 2001, subject to the legal availability of funds. At June 30, 1997 and December 31, 1996 cumulative undeclared, unpaid dividends relating to Series A-1 and A-2 Preferred Stock were accrued for financial reporting purposes in the amount of \$1,993,000 and \$1,828,000, respectively.

#### Class B Redeemable Preferred Stock

The Class B Redeemable Preferred Stock has a \$.01 par value, with 7,500 shares authorized, issued and outstanding. The Class B Redeemable Preferred Stock has a cumulative dividend payable semi-annually at June 30 and December 31. From July 1, 1995 through June 30, 1997, the dividend is calculated at a rate equal to 11.125% per annum of its liquidation value, and increases to 14.125% per annum thereafter. The Class B Redeemable Preferred Stock may be redeemed at its liquidation value together with all accrued and unpaid dividends at any time at the option of the Company. The liquidation preference of the preferred stock is the face amount, \$1,000 per share, plus all accrued and unpaid dividends. The Company is required to redeem all of the outstanding shares of the stock on December 31, 2001, subject to the legal availability of funds. At June 30, 1997 and December 31, 1996 cumulative undeclared, unpaid dividends relating to the Class B Redeemable Preferred Stock were accrued for financial reporting purposes in the amount of \$4,001,000 and \$3,587,000 respectively.

# 12% Cumulative Exchangeable Redeemable Preferred Stock

A maximum of 6,000,000 shares of 12% Cumulative Exchangeable Redeemable Preferred Stock, par value \$.01 per share, have been authorized for issuance. The Company has issued 3,595,586 shares of 12% Cumulative Exchangeable Redeemable Preferred Stock (the "Preferred Stock"). The Preferred Stock accrues a semi-annual dividend at the annual rate of 12% (\$1.20) per share, based on the liquidation preference of \$10 per share and is fully cumulative.

#### TELOS CORPORATION AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Through November 21, 1995, the Company had the option to pay dividends in additional shares of Preferred Stock in lieu of cash (provided there were no blocks on payment as further discussed below). Dividends are payable by the Company, provided the Company has legally available funds under Maryland law and is able to pay dividends under its charter and other corporate documents, when and if declared by the Board of Directors, commencing June 1, 1990, and on each six month anniversary thereof. Dividends in additional shares of the Preferred Stock were paid at the rate of 0.06 of a share for each \$.60 of such dividends not paid in cash. No dividends have been declared or paid during fiscal years 1992 through 1996. Cumulative undeclared dividends as of June 30, 1997 accrued for financial reporting purposes totaled \$12,578,000. Dividends for the years 1992 through 1994 and for the dividend payable June 1, 1995 were accrued under the assumption that the dividend will be paid in additional shares of preferred stock and are valued at \$3,950,000. Had the Company accrued these dividends on a cash basis, the total amount accrued would have been \$15,101,000. The dividends payable on December 1, 1995, June 1, 1996, December 1, 1996, and June 1, 1997 totaling \$8,628,000 were accrued on a cash basis.

The Company has not declared or paid dividends since 1991, due to restrictions and ambiguities relating to the payment of dividends contained within its charter, its working capital facility agreement, and under Maryland law.

Item 2.	Management's	Discussion	and Analysis	of	Financial	Condition	and
	Results of Op	erations.					

#### General

In the first six months of 1997, the Company had increased revenue and profitability as compared to 1996. The increased revenue resulted from increased order volume in the Systems Integration Group, as well as revenue generated from contracts awarded in 1997 to the Systems and Support Services Group. The increase in profitability was also attributable to the cost reductions and branch consolidation measures implemented by the Company in the last half of 1996. The profitability was further enhanced by an improvement in the product mix on the Company's long term contracts and improved profit margins realized on new contracts.

Total backlog from existing contracts was \$1.1 billion as of June 30, 1997 and is approximately the same as the total backlog as of December 31, 1996. As of June 30, 1997, the funded backlog of the Company totaled \$103 million, a decrease of \$12 million from December 31, 1996. Funded backlog represents aggregate contract revenues remaining to be earned by the Company at a given time, but only to the extent, in the case of government contracts, funded by a procuring government agency and allocated to the contracts.

#### Results of Operations

The condensed consolidated statements of income include the results of operations of Telos Corporation and its wholly owned subsidiaries Telos Corporation (California), Telos Field Engineering Inc. ("TFE"), enterWorks.com, inc. ("enterWorks"), and Telos International Corporation ("TIC"), ("the Company"). The major elements of the Company's operating expenses as a percentage of sales for the three and six month periods ended June 30, 1997 and 1996 are as follows:

	Three Mont June	ths Ended e 30,	June 30,	
	1997	1996 1997		1996
Sales	100.0%	100.0%	100.0%	100.0%
Cost of sales	83.5	87.5	84.6	88.4
SG&A expenses Goodwill amortization	11.8 0.4	15.4 0.6	11.9 0.4	15.3 0.6
	0.4	0.0	0.4	0.0
Operating income (loss)	4.3	(3.5)	3.1	(4.3)
Other income (expense)		(0.8)		(0.4)
Interest expense	(3.2)	(3.0)	(3.2)	(3.0)
Income tax provision				
Income (loss) from continuing operations	1.1	(7.3)	(0.1)	(7.7)
Discontinued operations:				
Income from discontinued operations		0.6		0.2
Net income (loss)	1.1%	(6.7)%	(0.1)%	(7.5)%
	===	===	===	===

The Company operates in two market segments: systems and support services (the "Systems and Support Services Group"), which consists of enterWorks and hardware and software support services; and the Systems Integration Group.

Sales, gross profit, and gross margin by market segment for the periods designated below are as follows:

Three Months Ended June 30,	Six Months Ended June 30,	
1997 1996		.996
(amounts in thousands	···· · · · · · · · · · · · · · · · · ·	
Revenues:		
	\$ 59,300 \$5	2,455
Systems Integration 24,907 17,643	53,134 3	1,574
Total \$58,089 \$43,866 \$		4,029
====== =====	======= =	=====
Gross Profit:		
Systems and Support Services \$5,886 \$4,130	\$9,653 \$	7,010
Systems Integration 3,692 1,384	7,622	2,764
Total 9,578 \$5,514	\$17,275 \$	9,774
	=====	=====
Gross Margin:		
Systems and Support Services 17.7% 15.8%	16.3%	13.4%
Systems Integration 14.8% 7.8%	14.3%	8.8%
Total 16.5% 12.6%	15.4%	11.6%

For the three month period ended June 30, 1997, revenue increased by \$14.2 million, or 32.4%, to \$58.1 million from \$43.9 million for the comparable 1996 period. The increase for the three month period includes a \$7.3 million increase in systems integration revenue and a \$6.9 million increase in systems and support services revenue.

The increase in the Systems Integration Group revenue of \$7.3 million results from increased orders on its Small Multiuser Computer II contract, as well as increased sales in other business lines of the division. The Company anticipates that the last half of 1997 will show stronger order and revenue volume. However, there can be no assurance that such order and sales volume growth will materialize.

The Systems and Support Services Group revenue increase of \$6.9 million is due to an increase in software support revenue of \$6.5 million, an increase in hardware support revenue of \$500,000, offset by a \$100,000 decrease in enterWorks revenue. The increase in software support is due to increased services under certain of the Company's large labor contracts. The increase is also attributable to the revenue recognized on its Immigration and Naturalization Service Blanket Purchase Agreement for Field Operation Support contract, which was awarded in early 1997. The increase in hardware support revenue is due to increased firm orders on its existing contracts, as well as sales generated on contracts awarded in 1997. The decline in enterWorks sales is due to a decline in order volume for the quarter. Revenue increased \$28.4 million or 33.8% to \$112.4 million for the six months ended June 30, 1997 from \$84.0 million for the comparable 1996 period. The increase for the six month period includes a \$21.6 million increase in systems integration revenue and a \$6.8 million increase in systems and support services revenue. The reasons for these revenue increases are discussed above.

Cost of sales increased by \$10.2 million or 26.5%, to \$48.5 million in the three month period ended June 30, 1997, from \$38.3 million in the comparable 1996 period. The increase in cost of sales for the three month period includes a \$5.0 million increase in systems integration and a \$5.2 million increase in systems and support services cost of sales. The increase in systems integration and system and support services cost of sales is due to the increased sales volume for 1997 as compared to 1996.

For the six months ended June 30, 1997, cost of sales increased \$20.9 million, or 28.1%, to \$95.2 million from \$74.3 million for the same period in 1996. The increase in cost of sales includes a \$16.7 million increase in systems integration cost of sales and a \$4.2 million increase in systems and support services cost of sales. The reasons for these cost of sales increases are discussed above.

Gross profit increased \$4.1 million in the three month period to \$9.6 million in 1997, from \$5.5 million in the comparable 1996 period. The increase in gross profit includes a \$2.3 million increase in systems integration gross profit, and a \$1.8 million increase in systems and support services gross profit. For the six month period, gross profit increased by \$7.5 million to \$17.3 million from \$9.8 million. This increase includes a \$4.9 million increase in systems integration gross profit and a \$2.6 million increase in systems and support services gross profit. The reasons for the gross profit increase for the periods ended June 30, 1997 compared to June 30, 1996 related to the changes in revenues and cost of sales for the respective periods. In addition, the Systems Integration Group experienced shifts in product mix on its large contracts which improved profit margins. The Systems and Support Services Group improved its gross margin by maximizing its efforts on profitable contracts and progressively reducing the number of less profitable contracts. The Group further enhanced its gross profit through cost reduction measures implemented in the fourth quarter of 1996.

Gross margins were 16.5% and 15.4%, respectively, for the three and six month periods of 1997 as compared to 12.6% and 11.6%, respectively, for the comparable periods of 1996.

Selling, general, and administrative expense ("SG&A") increased by approximately 100,000 or 1.5%, to \$6.9 million in the second quarter of 1997 from \$6.8 million in the comparable period of 1996. For the six month period of 1997, SG&A increased \$573,000 to \$13.4 million from \$12.8 million in 1996. These increases are primarily due to increased spending and investment by the Company on its enterWorks subsidiary. This investment was offset by a decline in bid and proposal and sales and marketing expense in both the Systems Integration Group and the rest of the Systems and Support Services Group as a result of the cost reductions and branch consolidation measures implemented by the Company in the last half of 1996. SG&A as a percentage of revenues decreased to 11.8% for the second quarter of 1997 from 15.4% in the comparable 1996 period. SG&A as a percentage of revenues for the six month period ended June 30, 1997 decreased to 11.9% from 15.3% compared to the same period in 1996.

Goodwill amortization expense decreased \$66,000 to \$209,000 for the three months and decreased by \$116,000 to \$434,000 for the six months ended June 30, 1997. These reductions are due to adjustments in the goodwill balance as a result of realization of acquired tax benefits resulting from the 1992 acquisition of Telos Corporation (California) and a writedown in the goodwill balance from the sale of TCS in 1996.

Operating income increased by \$4.0 million to a \$2.5 million operating profit in the 1997 three month period from \$1.5 million of operating loss in the comparable 1996 period. Operating income increased \$7.0 million to a \$3.4 million operating profit from a \$3.6 million operating loss for the six month period ended June 30, 1997 compared to the six month period ended June 30, 1996. These increases resulted from the aforementioned increases in gross profit.

Non operating income (expense) in the three and six month periods ended June 30, 1997 increased over the comparable 1996 periods due to the \$355,000 Rosecliff litigation settlement provision expense that the Company recorded in the second quarter of 1996.

Interest expense increased approximately \$547,000 to \$1.9 million in the second quarter of 1997 period from \$1.3 million in the comparable 1996 period, and increased approximately \$1.1 million to \$3.6 million for the six months ended June 30, 1997 from \$2.5 million for the comparable 1996 period. These increases are due to increased debt levels in 1997 as well as an increase in interest recorded for capital lease payments for leases entered into after March 1996.

The Company did not have an income tax provision for the three month and six month periods ended June 30, 1997 or 1996 due to its cumulative net losses in the six month 1997 and 1996 periods.

# Liquidity and Capital Resources

For the six months ended June 30, 1997, the Company used \$16.9 million of cash in its operating activities primarily as a result of a significant reduction in trade accounts payable and other Company obligations. The use of cash was also a result of a significant investment by the Company in its enterWorks division. The Company funded its net loss and use of operating cash as well as its investing activities through increased borrowings under its term facility.

As a result of the Company's sale of its TCS division for \$31.6 million in December 1996, the Company's short-term liquidity constraints have improved. However, the Company continues to aggressively manage its cash and reduce its discretionary spending. The Company also continues to evaluate its cost reduction programs and its investment in enterWorks.

At June 30, 1997, the Company had outstanding debt of \$50.0 million, consisting of \$33.4 million under the secured senior credit facility and \$16.6 million in subordinated debt. Subsequent to December 31, 1996, the Company's bank entered into an agreement with the Company to refinance its \$45 million term facility and extend the maturity date to July 1, 2000. The terms and conditions of the new facility are similar to the previous senior credit facility except for amendments made to certain of the financial and non financial covenants.

The Company is actively reviewing its financing requirements for enterWorks, and continues to fund on-going product development, sales and marketing, and business activities of the subsidiary. The Company will continue to evaluate various financing alternatives to maintain the enterWorks operations.

The Company continually evaluates its financing requirements to support its business base and anticipated growth. The Company anticipates that its current facility will be adequate for 1997. However, should faster than anticipated growth occur, the Company believes that an expanded senior credit facility would be required through a multi-bank syndication arrangement. Item 3. Defaults Upon Senior Securities

Senior and Class B Redeemable Preferred Stocks

The Company has not declared dividends on its Senior Redeemable Preferred Stock, Series A-1 and A-2, and its Class B Redeemable Preferred Stock since their issuance. Total undeclared unpaid dividends accrued for financial reporting purposes are \$1,993,000 for the Series A-1 and A-2 Preferred Stock and \$4,001,000 for the Class B Preferred Stock at June 30, 1997.

## 12% Cumulative Exchangeable Redeemable Preferred Stock

Through November 21, 1995, the Company had the option to pay dividends in additional shares of Preferred Stock in lieu of cash (provided there were no blocks on payment as further discussed below). Dividends are payable by the Company, provided the Company has legally available funds under Maryland law and is able to pay dividends under its charter and other corporate documents, when and if declared by the Board of Directors, commencing June 1, 1990, and on each six month anniversary thereof. Dividends in additional shares of the Preferred Stock were paid at the rate of 0.06 of a share for each \$.60 of such dividends not paid in cash. No dividends have been declared or paid during fiscal years 1992 through 1996. Cumulative undeclared dividends as of June 30, 1997 accrued for financial reporting purposes totaled \$12,578,000. Dividends for the years 1992 through 1994 and for the dividend payable June 1, 1995 were accrued under the assumption that the dividend will be paid in additional shares of preferred stock and are valued at \$3,950,000. Had the Company accrued these dividends on a cash basis, the total amount accrued would have been \$15,101,000. The dividends payable on December 1, 1995, June 1, 1996, December 1, 1996, and June 1, 1997 totaling \$8,628,000 were accrued on a cash basis.

The Company has not declared or paid dividends since 1991, due to restrictions and ambiguities relating to the payment of dividends contained within its charter, its working capital facility agreement, and under Maryland law.

# Item 4. Submission of Matters to a Vote of Security Holders

On May 24, 1997 at the annual meeting of common shareholders a vote was taken to elect the following directors: Dr. Fred Charles Ikle', John B. Wood, Norman P. Byers, and Dr. Stephen Bryen. The persons nominated were approved to be directors of the Corporation by unanimous vote of all shareholders present at the meeting which represented a majority of the Company's common shares outstanding.

Item 6. Exhibits and Reports on Form 8-K

- (a) Exhibits:
- 10.82 Amended and Restated Credit Agreement among Telos Corporation, a Maryland Corporation; Telos Corporation, a California Corporation; and NationsBank, N.A. dated as of July 1, 1997
- 27 Financial Data Schedule
- (b) Reports on Form 8-K:

None.

# SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

DATE:

Telos Corporation

August 14, 1997

/s/ Lorenzo Tellez Lorenzo Tellez (Principal Financial Officer & Principal Accounting Officer)

# Telos Corporation Exhibit Index

Exhibit Number Exhibit Name

10.82 Amended and Restated Credit Agreement among Telos Corporation; a Maryland Corporation; Telos Corporation; a California Corporation, and NationsBank, N.A. dated as of July 1, 1997

27 Financial Data Schedule

# AMENDED AND RESTATED CREDIT AGREEMENT

# AMONG

TELOS CORPORATION, A MARYLAND CORPORATION,

TELOS CORPORATION, A CALIFORNIA CORPORATION (COLLECTIVELY, BORROWERS)

AND

NATIONSBANK, N.A. (BANK)

NATIONSBANK, N.A. (AGENT)

-----

# Dated as of July 1, 1997

# TABLE OF CONTENTS

Page

ARTICLE I DEFINITIONS
ARTICLE II AMOUNT AND TERMS OF THE LOANS
ARTICLE III LETTERS OF CREDIT
ARTICLE IV REPRESENTATIONS AND WARRANTIES
ARTICLE V CONDITIONS OF EFFECTIVENESS OF THIS AGREEMENT AND LENDING
ARTICLE VI AFFIRMATIVE COVENANTS
ARTICLE VII NEGATIVE COVENANTS
ARTICLE VIII FINANCIAL COVENANTS
ARTICLE IX DEFAULTS; REMEDIES
ARTICLE X SETOFFS, ETC
ARTICLE XI EXPENSES
ARTICLE XII AMENDMENTS AND WAIVERS, ETC72
ARTICLE XIII PARTICIPATIONS
ARTICLE XIV INDEMNIFICATION
ARTICLE XV THE AGENT
ARTICLE XVI MISCELLANEOUS

THIS AMENDED AND RESTATED CREDIT AGREEMENT (this "Agreement") is made and entered into as of the \_\_\_\_\_\_ day of July, 1997, by and among (i) Telos Corporation, a Maryland corporation, formerly known as C3, Inc. ("Telos Corp.") and Telos Corporation, a California corporation ("Telos Sub") (individually, "Borrower" and collectively, "Borrowers"), and (ii) NationsBank, N.A., a national banking association, successor by merger to American Security Bank, N.A. (together with its successors and assigns, "NationsBank"). NationsBank in its capacity as Agent is sometimes referred to herein as the "Agent."

#### WITNESSETH:

A. Telos Corp., Telos Sub, NationsBank, and the Agent entered into that certain Revolving and Reducing Senior Facility Credit Agreement dated as of January 14, 1992, as subsequently amended through the date hereof (as amended, the "Original Credit Agreement").

B. Telos Corp., Telos Sub, NationsBank and the Agent desire to amend and restate the Original Credit Agreement in its entirety (i) to amend certain terms and provisions of the Original Credit Agreement, (ii) to reflect the amendments previously executed and delivered by Borrowers, NationsBank and the Agent and (iii) to make certain other changes.

C. The Banks (as defined in Article I) are willing, subject to the terms and conditions of this Agreement, to make available to Borrowers, on a revolving basis, up to a maximum principal amount of \$45,000,000.

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties, intending to be legally bound, hereby amend and restate the Original Credit Agreement in its entirety as follows:

#### ARTICLE I

#### DEFINITIONS

1.1 General Interpretive Principles. If the context requires, the use of any gender shall also refer to any other gender. Defined terms shall also mean in the singular number the plural and in the plural the singular. All accounting terms not specifically defined have the meanings determined by reference to United States generally accepted accounting principles, consistently applied ("GAAP"). The word "including" is not exclusive; if exclusion is intended, the word "comprising" is used instead. The word "or" shall mean "and/or" unless the context clearly prohibits that construction. All references to Articles, Sections, Exhibits and Schedules are to articles and sections of, and exhibits and schedules to, this Agreement.

1.2 Certain Definitions. As used herein, the following terms, when capitalized, have the following respective meanings:

Acceptance Date: The date on which customers of a Borrower or a Subsidiary become contractually obligated to pay for products delivered by each such Borrower or each such Subsidiary pursuant to the terms of the relevant contract (e.g., in the case of certain Military Departments under U.S. Government Contracts, the date of execution of DD Form 250, "Material Inspection and Receiving Report").

Accountants: The independent certified public accounting firm of recognized national standing selected by Telos Corp. and acceptable to the Agent.

#### Administrative Fee: The meaning specified in Section 2.5(d).

Affiliate: As applied to any Person, any member, director or officer of such Person, any corporation, association, firm or other entity of which such Person is a member, director or officer, and any other Person directly or indirectly controlling, controlled by or under direct or indirect common control with such Person. The term "control" means the possession, directly or indirectly, of the power to direct or to cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract, or otherwise; provided, however, that neither the Banks nor the Agent shall be considered an Affiliate of either Borrower or a Subsidiary for purposes of this Agreement.

#### Agent: NationsBank, and any successor agent.

Agreement: This Amended and Restated Credit Agreement and the Exhibits and Schedules attached hereto (all of which Exhibits and Schedules are hereby incorporated by reference and made a part hereof), as amended or supplemented from time to time in accordance with the terms hereof.

Application: The meaning specified in Section 3.1.

Assignment of Claims Act: 31 U.S.C. ss. 3727 and 41 U.S.C. ss. 15, and any similar or successor federal statute, and the rules and regulations issued thereunder, all as the same shall be in effect from time to time.

Assignment of Tenant's Interest Under Leases: The collateral assignments of the leases of Borrowers, all of which assignments shall be in proper recordable form as required by the laws of the applicable jurisdictions and otherwise in the form previously delivered to the Agent.

Audit Fee: The meaning specified in Section 2.5(b).

Bank or Banks: NationsBank or any other bank or financial institution which is a participant in, or assignee of, all or a portion of the Commitment, and their respective successors and assigns.

Banking Day: A day other than a Saturday, Sunday or other day on which commercial banks in Washington, D.C. or New York, New York are authorized or required by law to close.

Base Applicable Margin: The meaning specified in Section 2.4(b).

Base Interest Rate: The Base Rate plus the Base Applicable Margin.

Base Eurodollar Rate: For any Eurodollar Loan for any Interest Period therefor, the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) appearing on Telerate Page 3750 (or any successor page) as the London interbank offered rate for deposits in dollars at approximately 11:00 a.m. (London time) two (2) Banking Days prior to the first day of such Interest Period for a term comparable to such Interest Period. If for any reason such rate is not available, the term "Eurodollar Rate" shall mean, for any Eurodollar Loan for any Interest Period therefor, the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) appearing on Reuters Screen LIBO Page as the London interbank offered rate for deposit in dollars at approximately 11:00 a.m. (London time) two (2) Banking Days prior to the first day of such Interest Period for a term comparable to such Interest Period; provided, however, if more than one rate is specified on Reuters Screen LIBO Page, the applicable rate shall be the arithmetic mean of all such rates.

Base Rate: The rate of interest from time to time established and publicly announced by NationsBank, in its sole discretion, as its base (or prime) rate of interest to be used as an index in determining actual rates of interest to be charged to certain of NationsBank's borrowers. The Base Rate may not be the lowest rate charged by NationsBank to its borrowers. The rate shall be adjusted as of the effective date of each change in NationsBank's Base Rate.

Base Rate Loan: Any Loan for which interest on all or a portion of the outstanding principal thereof is to be computed with reference to the Base Interest Rate.

Borrowers: Telos Corp. and Telos Sub, or either of them as the context requires, and their respective successors and assigns.

Borrowing Base: At any time, the collateral value of certain assets of Borrowers and the Subsidiaries, on a consolidated basis, against which Loans shall be made by the Banks hereunder, calculated as provided in Section 2.2, as reflected in the most recently delivered Borrowing Base Certificate.

Borrowing Base Certificate: A certificate of the President, Vice President or the chief financial officer of Telos Corp., substantially in the form of Exhibit F hereto.

Capital Expenditure: An expenditure by a Borrower or a Subsidiary for property classified as a "fixed asset" under GAAP. Such term shall also include, whether or not required by GAAP, the obligations under a Capitalized Lease and expenditures for spare parts for Telos Field Engineering, Inc., but shall exclude expenditures specifically reimbursable under contracts.

Capitalized Lease: Any lease of property (real, personal or mixed) which, in accordance with GAAP, should be capitalized on the lessee's balance sheet or for which the amount of the asset and liability thereunder if so capitalized should be disclosed in a note to such balance sheet. For purposes of this Agreement, any Person shall be deemed to be the owner of any property which it has acquired or holds subject to a Capitalized Lease or conditional sale agreement or other arrangement pursuant to which title to the property has been retained by or vested in some other person for security purposes.

Change of Control: If John R.C. Porter or entities wholly-owned and controlled by John R.C. Porter shall no longer own or control, by contract, proxy or otherwise, an aggregate amount of at least 51% of the issued and outstanding voting stock of Telos Corp.

Closing Date: July 1, 1997.

Code: The Internal Revenue Code of 1986, as amended from time to time, and any Treasury regulations, revenue rulings or technical information releases issued thereunder.

Collateral: (i) The "Collateral" as such term is defined in the Security Agreements, (ii) the "Pledged Collateral" as such term is defined in the Pledge Agreements, and (iii) all other property, now owned or hereafter acquired, of a Borrower or a Subsidiary in which such Borrower or Subsidiary shall have created a security interest to secure the Loans, and all Proceeds thereof.

Collateral Account: The meaning specified in Section 2.15.

Collections: The meaning specified in Section 2.14.

Commission: The Securities and Exchange Commission and any other similar or successor agency of the federal government administering the Securities Act, the Exchange Act or the Trust Indenture Act.

Commitment: The Commitment of the Banks as set forth in Section 2.1(a).

Commitment Fee: The meaning specified in Section 2.5(a).

Compliance Certificate: The meaning specified in Section 6.8.

Consents: The meaning specified in Section 4.17.

Debt Ratio: The meaning specified in Section 2.4(d).

Deposit Accounts: The deposit accounts established by Borrowers with the Agent pursuant to Section 6.15 hereof.

Early Termination Fee: The meaning specified in Section 2.6.

EBITDA: For any period, consolidated net income of Borrowers and Subsidiaries before deductions for interest, taxes, depreciation and amortization, all as determined in accordance with GAAP.

Eligible Billed Commercial Receivables: All Receivables (excluding Eligible Billed Government Receivables) of Borrowers and the Subsidiaries (i) which arose from services performed or products delivered by a Borrower or a Subsidiary and such services have been performed or such products have been delivered for the appropriate obligor, (ii) which are based upon an enforceable order or contract, written or oral, for services performed or products delivered, (iii) the title to such Receivables is absolute and not subject to any prior Lien except Permitted Liens, (iv) in which the Agent has a first priority perfected security interest, (v) the obligors with respect to which are independent third parties having their principal places of business located in the United States of America, (vi) the Acceptance Date for which has occurred, (vii) for which the obligors have not notified any Borrower or Subsidiary of any dispute concerning, or claimed nonconformity of, any of the services or products from the sale of which such Receivables arose, (viii) which are due and payable not more than thirty (30) days from the original invoice date, (ix) which are not outstanding more than ninety (90) days after (A) for products, the later of the Acceptance Date or the original invoice date or (B) for services, the original invoice date, (x) which have not had any payment date extended, (xi) which do not arise out of any contract or order that, by its terms, forbids or makes void or unenforceable the assignment by such Borrower or Subsidiary to the Agent of such Receivable, (xii) for which the Borrowers and the Subsidiaries have not received any note, trade acceptance draft or other instrument, unless, if any such instrument has been received, Borrowers immediately notified the Agent and, at the Agent's request, endorsed or assigned and delivered such instrument to the Agent, (xiii) the obligors of which are not subsidiaries or other affiliates of any Borrower or Subsidiary, (xiv) which are not for advance payment on any order or contract (xv) which are in connection with "ship in place" or "bill and hold" practices or instructions or any other practices or instructions to hold services or products for any period of time, (xvi) the amount of which is reduced by the amount of any deposit received by such Borrower or Subsidiary, (xvii) which are not obligations of an obligor which has dissolved, terminated its existence or has become insolvent or is subject to any proceeding under bankruptcy or insolvency laws (however evidenced), (xviii) which do not relate in whole or in part to services or products which have been rejected by any obligor, (xix) which have not been reserved as doubtful and have not otherwise failed to meet credit standards acceptable to the Agent, (xx) which are not subject to any claim of reduction, counterclaim, setoff, recoupment, or any claim for credits, allowances, or a adjustments by the obligor for unsatisfactory services or for any other reason; (xxi) which the Agent has not deemed ineligible because of uncertainty as to the obligor or because the Agent otherwise reasonably considers the collateral value thereof to be impaired or its ability to realize such value to be insecure, and (xxii) which do not constitute a final billing or close out billing related to a completed contract; provided, however, that in the case of any obligor obligated in respect of Receivables having an aggregate face amount of \$500,000 or more, if more than fifty percent (50%) in aggregate amount of the Receivable of such obligor shall fail to meet any of the foregoing criteria, no Receivable of such obligor shall be an Eligible Billed Commercial Receivable as defined hereunder; provided, further, that no Receivable due to Telos International Corp., to Kuwait International Inc. or to any joint venture to which a Borrower or a Subsidiary is a party shall be included for purposes of this definition.

Eligible Billed Government Receivables: All Receivables arising out of U.S. Government Contracts of Borrowers and the Subsidiaries (i) which arose from services performed or products delivered by a Borrower or a Subsidiary and such services have been performed or such products have been delivered for the appropriate obligor, (ii) which are based upon an enforceable order or contract, written or oral, for services performed or products delivered, (iii) the title to such Receivables is absolute is and not subject to any prior Lien except Permitted Liens, (iv) in which the Agent has a first priority perfected security interest, (v) the Acceptance Date for which has occurred, (vi) for which the obligors have not notified any Borrower or Subsidiary of any dispute concerning, or claimed nonconformity of, any of the services or products from the sale of which such Receivables arose, (vii) which are due and payable not more than thirty (30) days from the original invoice date, (viii) which are not outstanding more than ninety (90) days after (A) for products, the later of the Acceptance Date or the original invoice date or (B) for services, the original invoice date, (ix) which have not had any payment date extended, (x) which do not arise out of any contract or order that, by its terms, forbids or makes void or unenforceable the assignment by such Borrower or Subsidiary to the Agent of such Receivable, (xi) for which the Borrowers and the Subsidiaries have not received any note, trade acceptance draft or other instrument, unless, if any such instrument has been received, Borrowers immediately notified the Agent and, at the Agent's request, endorsed or assigned and delivered such instrument to the Agent, (xii) which do not relate in whole or in part to services or products which have been rejected by any obligor, (xiii) which have not been reserved as doubtful and have not otherwise failed to meet credit standards acceptable to the Agent, (xiv) which is not subject to any claim of reduction, counterclaim, setoff, recoupment, or any claim for credits, allowances, or adjustments by the obligor for unsatisfactory services or for any other reason, (xv) which are not for advance payment on any order or contract, (xvi) which are in connection with "ship in place" or "bill and hold" practices or instructions or any other practices or instructions to hold services or products for any period of time, (xvii) the amount of which is reduced by the amount of any deposit received by such Borrower or Subsidiary and (xviii) which do not constitute a final billing or close out billing related to a completed contract; provided, however, that in the case of any Government Contract having an aggregate face amount of \$500,000 or more, if more than fifty percent (50%) of the Receivables under such Government Contract shall fail to meet any of the foregoing criteria, no Receivables of such Government Contract shall be an Eligible Billed Government Receivable as defined hereunder; provided, further, that no Receivable due to Telos International Corp., to Kuwait International Inc. or to any joint venture to which a Borrower or a Subsidiary is a party shall be included for purposes of this definition.

Eligible Hardware Inventory: Hardware Inventory of Borrowers (i) which has been purchased in connection with firm orders or created in the ordinary course of business, (ii) which can be shipped to a customer without repair or modification, except assembly or integration of components performed in the ordinary course of business, and (iii) in which the Agent has a first priority perfected security interest.

Eligible Receivables: Eligible Billed Government Receivables, Eligible Billed Commercial Receivables and Eligible Unbilled Government Receivables.

Eligible Unbilled Government Receivables: All unbilled Receivables of Borrowers and the Subsidiaries arising from Government Contracts (which by their terms are billable within thirty (30) days from the date on which the services were performed or the products delivered) pending billing (i) in which the Agent has a first priority perfected security interest and (ii) (x) which have remained unbilled for no more than such thirty (30) day period, (y) which will be billed within thirty (30) days of the date of the Borrowing Base Certificate prepared in respect of the period in which the services were performed or the products were delivered, and (z) which are due to be paid within thirty (30) days of such billing; provided, however, that no Receivable due to Telos International Corp., to Kuwait International Inc. or to any joint venture to which a Borrower or a Subsidiary is a party shall be included for purposes of this definition.

Enforcement Costs: All reasonable expenses, charges, costs and fees (including, without limitation, reasonable attorneys' fees and expenses) of any nature whatsoever paid or incurred by or on behalf of the Banks in connection with (a) the collection or enforcement of any or all of the Obligations, and/or any of the Revolving Loan Documents, and (b) the creation, perfection, collection, maintenance, preservation, defense, protection, realization upon, disposition, collection, sale or enforcement of all or any part of the Collateral.

enterWorks: enterWorks.com, Inc., a Delaware corporation, and its successors and assigns.

enterWorks Notes: Collectively, those certain Series A Senior Subordinated Unsecured Notes Due July 11, 2001 dated July 12, 1996 issued by enterWorks.

enterWorks Stock: The capital stock of, and any other equity or ownership interests in, enterWorks and all certificates representing such stock, equity or ownership interests, and all dividends, cash, instruments and other property and

rights received, receivable or distributed in respect of or in exchange for such stock, equity or ownership interests.

enterWorks Subordination Agreement: That certain Subordination Agreement dated as of July 12, 1996 among NationsBank, Telos Corp., enterWorks and each of the holders of the enterWorks Notes.

Equipment: The meaning specified in the U.C.C.

ERISA: The Employee Retirement Income Security Act of 1974, as amended from time to time, and the regulations issued pursuant to its authority.

ERISA Affiliate: All members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control which, together with Borrowers and the Subsidiaries, are treated as a single employer under Section 414(b), 414(c) or 414(m) of the Code.

ERISA Termination Event: (a) A "Reportable Event" described in Section 4043 of ERISA and the regulations issued thereunder (other than a "Reportable Event" not subject to the provision for 30-day notice to the Pension Benefit Guaranty Corporation ("PBGC") under such regulations), (b) the withdrawal of a Borrower, a Subsidiary or any ERISA Affiliate from a Plan during a Plan year in which it was a "substantial employer" as defined in Section 4001(a)(2) of ERISA, (c) the filing of a notice of intent to terminate a Plan or the treatment of a Plan Amendment as a termination under Section 4041 of ERISA, (d) the institution of proceedings to terminate a Plan by the PBGC, or (e) any other event or condition which might constitute grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan.

Eurodollar Applicable Margin: The meaning specified in Section 2.4(g).

Eurodollar Banking Day: A Banking Day on which commercial banks are open for international business (including dealings in U.S. dollar deposits) in London.

Eurodollar Loan: Any Loans for which interest on all or a portion of the outstanding principal thereof is to be computed with reference to the Eurodollar Rate.

Eurodollar Rate: The Base Eurodollar Rate plus the Eurodollar Applicable Margin specified in Section 2.4(g).

Event of Default: The meaning specified in Section 9.1.

Excepted Collateral: The Property, Plant and Equipment.

Exchange Act: The Securities Exchange Act of 1934, and any similar or successor federal statute, and the rules and regulations of the Commission thereunder, all as the same shall be in effect from time to time.

Excluded Claims: The meaning specified in Section 14.1.

Executive Officer: The meaning specified in Section 5.1(d).

Federal Reserve Board: The Board of Governors of the Federal Reserve System.

Financial Statements: The meaning specified in Section 4.7.

Franchises: The meaning specified in Section 4.11.

Funded Debt: As of the date of determination, the sum of Loan Outstandings, outstanding Subordinated Debt and the outstanding amount under the Capitalized Leases.

Government Contracts: Any contract with the United States or with any state or political subdivision thereof, or any department, agency or instrumentality of the United States, or any state or political subdivision thereof.

Guarantee: By any Person, any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Indebtedness or obligation of any other Person in any manner, whether directly or indirectly, and including, without limitation, any obligation of such Person, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or to purchase (or to advance or supply funds for the purchase of) any security for the payment of such Indebtedness, (ii) to purchase property, securities or services for the purpose of assuring the owner of such Indebtedness of the payment of such Indebtedness, or (iii) to maintain working capital, equity capital or other financial statement condition of the primary obligor so as to enable the primary obligor to pay such Indebtedness; provided, however, that the term "Guarantee" shall not include endorsements for collection or deposit in the ordinary course of business or obligations in respect of performance or bid bonds granted in the ordinary course of business and not to exceed \$250,000 in the aggregate.

Guaranties of Subsidiaries: The Guaranties of the Subsidiaries in the form

attached hereto as Exhibit B.

Indebtedness: With respect to any Person, (i) all obligations of such Person for borrowed money, or with respect to deposits or advances of any kind, (ii) all obligations of such Person evidenced by loan agreements, bonds, debentures, notes or similar instruments, (iii) all obligations of such Persons upon which interest charges are customarily paid, (iv) all obligations of such Person issued or assumed as the deferred purchase price of property or services (other than accounts payable to suppliers incurred in the ordinary course of business), (v) all Indebtedness of others secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien on property owned or acquired by such Person whether or not the obligations secured thereby have been assumed, (vi) all obligations of such Person under Capitalized Leases, (vii) all Guarantees of such Person, and (viii) all reimbursement obligations of such Person (whether contingent or otherwise) in respect of letters of credit, time drafts or similar obligations.

Indemnified Party: The meaning specified in Section 14.1.

Interest Period: (a) As to any Eurodollar Loan, the period commencing on and including the date of such Loan (or on the effective date of the election pursuant to Section 2.4(e) by which such Loan became a Eurodollar Loan) and ending on and including the day preceding the same day (or if there is no such same day, the date preceding the last day) in the first, third, or sixth calendar month thereafter, as selected by Borrowers in accordance with Section 2.4(e), and thereafter such period commencing on and including the day immediately following the last day of the then ending Interest Period for such Loan and ending on and including the day preceding the day corresponding to the first day of such Interest Period (or if there is no such corresponding day, the day preceding the last day), in the first, third, or sixth calendar month thereafter, as so selected by Borrowers; provided, however, that if any such Interest Period would otherwise end on a day prior to a day that is not a Banking Day, it shall be extended so as to end on the day prior to the next succeeding Banking Day unless the same would fall in a different calendar month, in which case such Interest Period shall end on the day preceding the first Banking Day preceding such next succeeding Banking Day; and (b) as to any Base Rate Loan, the period commencing on and including the date of such Loan (or on the effective date of the election pursuant to Section 2.4(e) by which such Loan became a Base Rate Loan) and continuing until terminated by the Borrowers pursuant to a subsequent election pursuant to Section 2.4(e) provided, however, that if any such Interest Period would otherwise end on a day prior to a day that is not a Banking day, it shall be extended so as to end on the day prior to the next succeeding Banking Day.

Interest Rate Protection Agreement: Any interest rate swap agreement, interest rate cap agreement, interest rate collar agreement or other similar agreement or arrangement, executed in connection with this Agreement, designed to protect Borrowers against fluctuations in interest rates.

Interest Rate: The Base Interest Rate or the Eurodollar Rate, as applicable.

Interest Rate Request: The meaning specified in Section 2.4(j).

Inventory: The meaning assigned to such term in the U.C.C.

Investor Notes: Collectively, those certain Series B Senior Subordinated Secured Notes and those certain Series C Senior Subordinated Unsecured Notes and subordinated in right of payment in the manner and to the extent set forth in the Investor Subordination Agreement.

Investor Subordination Agreement: That certain Subordination Agreement, dated as of October 13, 1995, among NationsBank, Borrowers and each of the holders of the Investor Notes.

Issuance Effective Date: The meaning specified in Section 3.2.

Items of Payment: All checks, drafts, cash and other remittances in payment or on account of payment of the Receivables (as defined in the U.C.C.) and any other Collateral and the cash proceeds of any returned, rejected or repossessed goods, the sale or lease of which gave rise to a Receivable.

Leased Real Property: Real property leased by a Borrower or a Subsidiary (as lessee) at any time.

Legal Requirement: Any requirement imposed upon any bank by any law of the United States of America or by a regulation, order, interpretation, ruling or official directive (whether or not having the force of law) of the Federal Reserve Board, the Federal Deposit Insurance Corporation, or any other board or governmental or administrative agency of the United States of America or any political subdivision thereof.

Letter of Credit: The meaning specified in Section 3.1.

Lien: (a) Any interest in property (whether real, personal or mixed and whether tangible or intangible) which secures an obligation owed to, or a claim

by, a Person other than the owner of such property, whether such interest is based on common law, statute or contract, including without limitation, any such interest arising from a Capitalized Lease, arising from a mortgage, charge, pledge, security agreement, conditional sale, lease, trust receipt or deposit in trust, or arising from a consignment or bailment given for security purposes (other than a trust receipt or deposit given in the ordinary course of business which does not secure any obligation for borrowed money), (b) any encumbrance upon such property which does not secure such an obligation, and (c) any exception to any policy of title insurance for or defect in the title to or ownership interest in such property, including, without limitation, reservations, rights of entry, possibilities of reverter, encroachments, easements, rights of way, restrictive covenants and licenses.

Loan or Loans: The meanings specified in Section 2.1(a).

Loan Account: The meaning specified in Section 2.14.

Loan Outstandings: As of any date, the aggregate outstanding principal amount of the Loans.

Loan Request: The meaning specified in Section 2.1(b).

Material Agreements: The meaning specified in Section 4.12.

Maturity Date: The meaning specified in Section 2.1(d).

NationsBank: NationsBank, N.A., a national banking association, and its successors and assigns.

Net Book Value: Solely when applied to Inventory, the net book value of such Inventory, as determined in accordance with GAAP without giving effect to any adjustment made to state inventory on the basis of the last-in-first-out method.

Net Worth: Consolidated stockholder's equity (including the Telos Corp. Preferred Stock) of Borrowers and the Subsidiaries, less any amount reflected as treasury stock.

New Subsidiaries: Collectively, Telos Field Engineering, Inc., a Delaware corporation, Telos International Corp., a Delaware corporation, enterWorks, Kuwait International Inc., a Delaware corporation, and their respective successors and assigns.

Note or Notes: The meaning specified in Section 2.3.

Obligations: Collectively, (i) all payment obligations of Borrowers and the Subsidiaries to the Banks under the Revolving Loan Documents including, without limitation, all obligations of Borrowers and the Subsidiaries in respect of payment of the principal of and interest on the Loans and all commissions, fees, expenses and other amounts payable by Borrowers and the Subsidiaries under the Revolving Loan Documents and (ii) all extensions, renewals, modifications, and substitutions of (or for) any of the foregoing.

Periodic LC Fee: The meaning specified in Section 3.2.

Permitted Investments: (i) Securities issued that are directly and fully guaranteed or insured by the United States Government or any agency or instrumentality thereof having maturities of not more than six months from the date of acquisition, (ii) time deposits and certificates of deposit of NationsBank, or any other domestic commercial bank having capital and surplus in excess of \$100,000,000 (and the holding company of such other domestic commercial bank which has outstanding commercial paper meeting the requirements specified in clause (iv) below) having maturities of not more than six months from the date of acquisition, (iii) repurchase agreements with a term of not more than seven days for underlying securities of the types described in clauses (i) and (ii) above (provided that the underlying securities of the type described in clause (i) may have maturities of more than six months from the date of acquisition) entered into with any bank meeting the qualifications specified in clause (ii) above or with securities dealers of recognized national standing, provided that the terms of such agreements comply with the guidelines set forth in the Federal Financial Institutions Examination Council Supervisory Policy Repurchase Agreements of Depositary Institutions With Securities Dealers and Others as adopted by the Comptroller of the Currency on October 31, 1985 (the "Supervisory Policy"), and provided, further, that possession or control of the underlying securities is established as provided in the Supervisory Policy, and (iv) commercial paper rated (as of the date of acquisition thereof) at least A-1 or the equivalent thereof by Standard & Poor's Corporation and P-1 or the equivalent thereof by Moody's Investors Service, Inc. and in either case maturing within six months after the date of its acquisition. The purchase of forward contracts by a Borrower or a Subsidiary shall not be Permitted Investments.

Permitted Liens: (a) Liens for taxes, assessments or other governmental charges or claims the payment of which is not at the time required, (b) statutory Liens of landlords and Liens of carriers, warehousemen, mechanics and materialmen and other like liens incurred in the ordinary course of business for

sums not yet due, (c) Liens (other than any Lien imposed by ERISA) incurred or deposits made in the ordinary course of business in connection with workman's compensation, unemployment insurance and other types of social security, (d) Liens incurred in connection with the Indebtedness relating to Capital Expenditures permitted under Section 7.1(c) hereof, (e) such imperfections of title, covenants, restrictions, easements and encumbrances on real property which, in each case, do not arise out of the incurrence of any Indebtedness and which do not interfere with or impair in any material respect the utility or value of the real property on which such Lien is imposed, (f) purchase money security interests in the ordinary course of business for non-capital expenditures and (g) the possessory Liens created in favor of the Agent pursuant to the Security Documents.

Person: A corporation, an association, a partnership, a joint venture, an organization, a business, an individual, a government or political subdivision thereof or a governmental agency.

Plan: Any employee benefit plan as defined in Section 3(3) of ERISA that a Borrower or a Subsidiary maintains or to which a Borrower or a Subsidiary is obligated to contribute for the benefit of employees or former employees of a Borrower or a Subsidiary.

Pledge Agreements: Those certain Pledge Agreements among Borrowers and the Agent, substantially in the form of Exhibit D hereto, as further amended, modified or supplemented from time to time in accordance with the respective terms thereof.

Pledged Stock: The stock of Telos Sub and certain of the Subsidiaries which has been pledged to NationsBank pursuant to the Pledge Agreements.

Proceeds: The meaning assigned to such term in the U.C.C.

Property, Plant and Equipment: (i) Any real or personal property, plant, building, facility, structure, equipment or unit, or other asset owned, leased or operated by Telos Corp., (ii) all of Telos Corp.'s owned equipment and fixtures, including, without limitation, furniture, machinery, vehicles and trade fixtures, together with any and all accessions, parts and appurtenances thereto, substitutions therefor and replacements thereof owned, leased or operated by Telos Corp., and (iii) all of the proceeds of the property described in clauses (i) and (ii) of this definition.

Public Preferred Stock: Telos Corp.'s 12% Cumulative Exchangeable Redeemable Preferred Stock.

Receivables: All Accounts (as defined in the U.C.C.), of Borrowers and the Subsidiaries on a consolidated basis, whether now owned or hereafter acquired, whether billed or unbilled and whether now existing or hereafter arising, including, without limitation, those arising under the Government Contracts, including the products and Proceeds of any thereof.

Regulation G, Regulation T, Regulation U and Regulation X: Regulations G, T, U and X, respectively, of the Board of Governors of the Federal Reserve System from time to time in effect including any successor or other regulation or official interpretation of said Board of Governors relating to the extension of credit by banks for the purpose of purchasing or carrying margin stocks applicable to member banks of the Federal Reserve System or to the extension of credit by brokers or dealers or persons other than banks, brokers or dealers for the purpose of purchasing or carrying margin stocks applicable to persons who are required to register with said Board of Governors under Part 207, 220, 221 and 224, respectively, of Title 12 to the Code of Federal Regulations.

Revolving Loan Documents: This Agreement, the Notes, the Security Documents, the Investor Subordination Agreement, the enterWorks Subordination Agreement and any other documents, instruments, certificates or agreements executed or delivered, or required to be executed or delivered, pursuant to or in connection with any of the foregoing.

Securities Act: The Securities Act of 1933, as amended, and any similar or successor federal statute, and the rules and regulations of the Commission thereunder, all as the same shall be in effect from time to time.

Security Agreements: Those certain Security Agreements between each of the Borrowers and the Subsidiaries and the Agent, substantially in the form of Exhibit C hereto, as further amended, modified or supplemented from time to time in accordance with the terms thereof.

Security Documents: The Security Agreements, the Guaranties of Subsidiaries, the Pledge Agreements and the Assignment of Tenant's Interest under Leases.

Series B Senior Subordinated Secured Notes: The seven (7) Series B Senior Subordinated Secured Notes Due October 1, 2000 dated October 13, 1995 issued by Telos Corp.

Series C Senior Subordinated Unsecured Notes: The seven (7) Series C Senior Subordinated Notes Due October 1, 2000 dated October 13, 1995 issued by Telos Subordinated Debt: The debt evidenced by the Investor Notes and the enterWorks Notes and any amounts incurred to repay or refinance all or a portion thereof ("Refinancing Debt"), provided that (i) the scheduled principal payments on any Refinancing Debt do not occur prior to the scheduled principal payments on the Investor Notes or the enterWorks Notes, respectively, being refinanced by such Refinancing Debt and (ii) the Refinancing Debt is subordinated in right of payment to Senior Indebtedness (as defined in the Investor Subordination Agreement and the enterWorks Subordination Agreement) on the terms and conditions substantially similar to those set forth in the Investor Subordination Agreement and the enterWorks Subordination Agreement, as applicable.

Subordinated Loan Documents: The Investor Notes, the Subordinated Security Agreement, the enterWorks Notes, the Investor Subordination Agreement and the enterWorks Subordination Agreement and any other documents executed and delivered in connection with the Subordinated Debt.

Subordinated Security Agreement: That certain Collateral Trust and Security Agreement, dated as of February 1, 1996, among Telos Corp., State Street Bank and Trust Company, as trustee, and the holders of the Series B Senior Subordinated Secured Notes, as such agreement may be amended, modified or supplemented from time to time in accordance with the terms thereof.

Subsidiaries: Telos Computing, Inc., a California corporation, Telos Software Products, Inc., a California corporation, DMA, Inc., a Wisconsin corporation, the New Subsidiaries, and their respective successors and assigns, and any other subsidiary of either Borrower or Subsidiary.

Tangible Capital Funds: The sum of Net Worth and Subordinated Debt, minus all intangible assets of Borrowers and the Subsidiaries (including, without limitation, goodwill, organizational costs, unamortized bank debt, discounts and expenses, franchises, licenses, patents and trademarks).

Tax: Any tax, levy, impost, duty, deduction, withholding or other charges of whatever nature required by any Legal Requirement (i) to be paid by the Banks or (ii) to be withheld or deducted from any payment otherwise required hereby to be made by a Borrower or a Subsidiary to the Banks; provided, however, that the term "Tax" shall not include any taxes imposed upon the net income of any Bank.

Telerate Page 3750: The British Bankers Association Libor Rates (determined at 11:00 a.m. London time) that are published by Dow Jones Telerate, Inc.

Telos Corp.: Telos Corporation, a Maryland corporation, and its successors and assigns.

Telos Corp. Preferred Stock: The Public Preferred Stock and the UBS Preferred Stock.

Telos Sub: Telos Corporation, a California corporation, and its successors and assigns.

Total Liabilities: At any date, all liabilities of Borrowers and the Subsidiaries, on a consolidated basis, as shown on their most recent consolidated financial statements.

Treasury Obligation: The meaning specified in Section 2.4(g).

Trust Indenture Act: The Trust Indenture Act of 1939, and any similar or successor federal statute, and the rules and regulations of the Commission thereunder, all as the same may be in effect from time to time.

UBS: Union de Banques Suisses (Luxembourg) S.A, and its successors and assigns.

UBS Preferred Stock: The Class B Preferred Stock issued by Telos Corp. to . UBS on January 14, 1992.

U.C.C.: The Uniform Commercial Code as in effect in the Commonwealth of Virginia from time to time.

U.S. Government Contracts: Government Contracts with the United States of America.

Warrant Agreement: That certain Warrant Agreement, dated as of January 14, 1992, between Telos Corp. and UBS, pursuant to the terms of which Telos Corp. granted to UBS rights under certain circumstances to purchase certain percentages of Telos Corp. Class A common stock, as amended, modified or supplemented from time to time.

1.3 Use of "Subsidiary". With respect to any representation or warranty set forth in this Agreement, the term "Subsidiary" shall refer to each Subsidiary in existence at the time the representation or warranty is made or deemed to be made. Covenants and other provisions shall apply to Subsidiaries actually in existence from time to time.

Corp.

2.1 Loans.

(a) Establishment. Subject to the terms and conditions hereof, upon thesatisfaction of the conditions precedent set forth in Article V and in relianceon the representations and warranties contained herein, the Banks shall advance to Borrowers up to a maximum principal amount of \$45,000,000 (the "Commitment"). Each borrowing hereunder shall be referred to as a "Loan" and all borrowings made hereunder shall be referred to herein as the "Loans."

(b) Loan Requests. Subject to the provisions of Sections 2.1(a) and 2.2, each Loanhereunder shall be made by the Banks upon Borrowers' request. All requests (a"Loan Request") for a Loan hereunder shall be made to the Agent and shall be in an amount not less than \$100,000. Each Loan Request shall be made pro rata among the Banks and shall be funded pro rata among the Banks. Loans shall be made on aBanking Date at such times before the Maturity Date as Borrowers may request prior to 12:00 noon (Washington, D.C. time) on such Banking Day. Such LoanRequests (i) shall specify the proposed date and the aggregate amount of the requested Loans and (ii) shall be irrevocable. The Banks shall make Loans on such proposed date in United States Dollars in immediately available funds. All proceeds of the Loans shall be credited to one or more Deposit Accounts.

(c) Continuing Validity of Representations and Warranties. Each Loan Request shall be deemed to include a representation and warranty by Borrowers that all of the representations and warranties of Borrowers set forth in the Revolving Loan Documents continue to be correct and complete and that all conditions precedentto the requested Loan have been satisfied, as of the time the Loan Request issubmitted to the Agent.

(d) Maturity Date. Subject to the provisions of Sections 2.6 and 2.7, all loans shall mature on July 1, 2000 or on such other date on which the principal amount of the Notes becomes due and payable as herein provided, whether by declaration of acceleration, prepayment or otherwise (the "Maturity Date").

(e) Prepayment and Reborrowings. During the period prior to the Maturity Date, Borrowers may, at their option, from time to time prepay all or any portion of the Loans made from time to time hereunder, subject to the provisions of Section 2.6. Borrowers may reborrow amounts so repaid provided that such reborrowings do not exceed the Commitment, subject to the provisions of Sections 2.1(d) (with respect to the maturity date of such Loans), 2.2 and 2.7.

2.2 Limitation on Loans; Borrowing Base. Borrowers acknowledge that the Loans Outstanding shall at no time exceed the lesser of:

- (a) the Commitment, or
- (b) the Borrowing Base, which is equal to:
  - (i) ninety percent (90%) of the face amount of Eligible Billed Government Receivables;
  - (ii) eighty-five percent (85%) of the face amount of Eligible Billed Commercial Receivables;
  - (iii) the lesser of (A) fifty percent (50%) of Eligible Unbilled Government Receivables and (B) \$7,000,000; and
  - (iv) the lesser of (A) fifty percent (50%) of the aggregate Net Book Value of Eligible Hardware Inventory and (B) \$11,000,000

The Borrowing Base shall be determined by reference to Borrowers and the Subsidiaries on a consolidated basis.

2.3 Notes. The Loans of each Bank shall be evidenced by a promissory note of Borrowers made payable to such Bank in a maximum principal amount equal to the commitment of such Bank (individually, as amended and in effect from time to time, a "Note" and collectively, the "Notes"). On the Closing Date, Borrowers shall execute an Amended and Restated Note in the form attached hereto as Exhibit A. The outstanding principal amount of the Notes shall be finally due and payable on the Maturity Date. The Notes shall bear interest (computed on the basis of the actual number of days elapsed over a 360-day year) on the unpaid principal amount thereof at the applicable rate or rates set forth in this Article II.

# 2.4 Interest.

(a) Interest Rates. Until the Maturity Date, interest on the outstanding principal amount of the Loans shall accrue for each day at either the Base Interest Rate for such day or the Eurodollar Rate for the Interest Period which includes such day, all as elected and specified (including specification as to length of Interest Period, as permitted by the definition of that term, with respect to any election of the Eurodollar Rate) by Borrowers in accordance with Section 2.4(i) hereof.

(b) Base Rate Option. The outstanding principal amount of the Loans shall bear interest until paid in full at the Base Rate plus a margin (the "Base Applicable Margin") equal to 1.00% per annum, subject to adjustment as set forth in Sections 2.4(c) and (d).

(c) Increase in Base Applicable Margin. Notwithstanding the foregoing, the Base Applicable Margin shall be increased by one quarter of one percent (0.25%) in the event that on any day of the month for which interest is calculated, the Loan Outstandings, as calculated by the Agent, exceeds the sum of (i) ninety percent (90%) of the face amount of Eligible Billed Government Receivables and(ii) eighty-five percent (85%) of the face amount of Eligible Billed Commercial Receivables, as calculated by the Agent.

(d) Reduction in Base Applicable Margin. Borrowers shall be entitled to a reduction of the Base Applicable Margin in the event that and at such time that (i) Net Worth is determined by the Agent to be equal to \$12,000,000 or greater, and (ii) Borrowers have requested the Agent and the Banks in writing to permanently reduce the amount set forth in Section 2.2(b)(iii) (B) from \$7,000,000 to \$2,000,000. In such event, the amount of reduction of the Base Applicable Margin shall be determined quarterly based upon the calculation of the Debt Ratio as follows:

Debt Ratio	Amount of Reduction of Base Applicable Margin
If the Debt Ratio is less than 4:1	0.25%
If the Debt Ratio is less than 3:1	0.50%
If the Debt Ratio is less than 2:1	0.75%

For purposes of this Section 2.4(d), "Debt Ratio" shall mean the ratio of Funded Debt to EBITDA. For purposes of calculating the Debt Ratio, (i) Funded Debt shall be calculated as the average of month-end Funded Debt on a rolling twelve (12) month basis and (ii) EBITDA shall be calculated on a rolling twelve (12) month basis, based upon the four (4) most recent Form 10Q reports delivered by Borrowers to Agent.

The determination of reductions, if any, to the Base Applicable Margin shall commence on the date of receipt by the Agent of Borrowers' audited financial statements for fiscal year 1997 and shall be determined quarterly thereafter upon receipt by the Agent of Borrowers' Form 10Q reports.

(e) Determination of Interest Rates. Upon the request of Borrowers, the Agent shall, as soon as practicable, notify Borrowers of each determination of a Eurodollar Rate; provided, however, that any failure to do so shall not relieve Borrowers of any liability hereunder. Each determination of an Interest Rate by the Agent pursuant to any provision of this Agreement shall be conclusive and binding on Borrowers in the absence of manifest error. The Agent shall, at the request of Borrowers, deliver to Borrowers a statement showing the quotations used by the Agent in determining any Interest Rate pursuant hereto.

- (f) Payment of Interest.
  - (i) Interest accruing on the outstanding principal amount of any Base Rate Loan shall be due and payable monthly in arrears on the first day of the month following the month for which interest has accrued.
  - (ii) Interest accruing on the outstanding principal amount of any Eurodollar Loan during any Interest Period shall be jointly and severally payable by Borrowers on the last Banking Day of such then current Interest Period; provided, however, that

with respect to Eurodollar Loans for which

the Interest Period selected by Borrowers pursuant to Section 2.4(e) hereof is greater than three (3) months, interest shall be payable quarterly on the last Banking Day of such quarterly period with the first such quarterly period commencing on the first day of the applicable Interest Period with any remaining unpaid interest being due and payable on the last day of such Interest Period; provided, further,

that all accrued interest on any Eurodollar Loan converted or prepaid prior to the last Banking Day of the applicable Interest Period shall be paid immediately upon such prepayment or conversion.

(iii) All interest accrued on the Notes but unpaid as of the Maturity Date (whether by acceleration or otherwise) shall be payable on the Maturity Date. In no event shall the amount paid or agreed to be paid by Borrowers as interest on the Notes exceed the highest lawful rate permissible under any law applicable thereto.

(g) Eurodollar Rate Option. Notwithstanding the foregoing, upon Borrowers' election of a Eurodollar Rate pursuant to Section 2.4(i), the outstanding principal amount of the Eurodollar Loan shall bear interest until paid in full at the Base Eurodollar Rate plus a margin (the "Eurodollar Applicable Margin") according to the Base Interest Rate in effect at the time, as follows:

Base Interest Rate	Eurodollar Applicable Margin
Base Rate + 1.25%	+ 4.00%
Base Rate + 1.00%	+ 3.75%
Base Rate + 0.75%	+ 3.50%
Base Rate + 0.50%	+ 3.25%
Base Rate + 0.25%	+ 3.00%

(h) Inability to Determine Base Eurodollar Rate. In the event that the Agent shall have determined (which determination shall be conclusive and binding upon Borrowers) that, by reason of circumstances affecting the London interbank eurodollar market, adequate and reasonable means do not exist for ascertainingthe Base Eurodollar Rate for any requested Interest Period with respect to a Loan that Borrowers have requested be made or converted as a Eurodollar Loan, the Agent shall forthwith give notice of such determination to Borrowers at least one (1) day prior to the proposed date for funding or converting such Eurodollar Loan. If such notice is given, any requested Eurodollar Loan shall be made or converted as a Base Rate Loan. Until such notice has been withdrawn by the Agent, Borrowers will not request that any Loan be made as or converted to a Eurodollar Loan.

(i) Election of Interest Rates. By a proper and timely Interest Rate Request in accordance with Section 2.4(j), Borrowers shall select the initial Interest Rate to be charged on advances of the Loans. Not more frequently than one (1) time in each calendar month, Borrowers may elect in writing an initial Interest Rate for any advance of the Loans, or to convert the Interest Rate on any sums outstanding under the Loans to any other Interest Rate (including, when applicable, the selection of the Interest Period); provided, however, that;

- (i) Borrowers shall not select any Interest Period that extends beyond the Maturity Date;
- (ii) except as otherwise provided in Section 2.4(j), no such change from the Eurodollar Rate to the Base Interest Rate shall become effective on a day other than the day, which must be a Banking Day, and, if such change involves a Loan upon which interest is, or will be, calculated at the Eurodollar Rate, also a Eurodollar Banking Day, next following the last day of the Interest Period last in effect for such Eurodollar Loan;
- (iii) any change of the Interest Rate on a Loan made on a date on which principal of such Loan is scheduled to be paid shall be made only after

such payment is made;

- (iv) any elections made by Borrowers pursuant to this Section 2.4(i) with respect to a Eurodollar Loan shall be in the minimum amount of \$5,000,000, plus any additional increments of \$1,000,000;
- (v) the aggregate amount of all Eurodollar Loans shall not exceed \$20,000,000 at any one time; and
- (vi) the first day of each Interest Period as to a Eurodollar Loan shall be a Banking Day.

In the absence of an election by Borrowers of the Eurodollar Rate, or, having made such election but Borrowers fail or are not entitled under the terms of this Agreement to elect to continue such Interest Rate and specify the applicable Interest Period therefor, then upon the expiration of such then current Interest Period, interest on the Loan shall accrue for each day at the Base Interest Rate for such day, until Borrowers, in accordance with this Section 2.4(i), elect a different Interest Rate and specify the Interest Period for the Loan.

(j) Interest Rate Request. The Agent will not be obligated to make Loans, convert the Interest Rate on the Loans to another Interest Rate or to act upon any election by Borrowers pertaining to Interest Rates or Interest Periods unless the Agent shall have received an irrevocable written notice (an "Interest Rate Request") from Borrowers at the times and specifying the information as follows:

- (a) the amount to be borrowed, prepaid or converted;
- (b) any election among the Base Interest Rate or the Eurodollar Rate;
- (c) the requested date on which such election is to be effective; and
- (d) the length of the Interest Period applicable to such advance of the Loan.

Such Interest Rate Request (or telephone advice thereof promptly confirmed in writing) shall be received by the Agent prior to 12:00 p.m. Baltimore, Maryland time, at least:

- three (3) Eurodollar Banking Days prior to the requested effective date of such election in the case of Eurodollar Loans,
- (ii) on the effective date of such election in the case of Base Rate Loans.

(k) Indemnity. Borrowers jointly and severally agree to indemnify and reimburse the Agent and the Banks and to hold the Agent and the Banks harmless from any loss, cost (including administrative costs) or expense which the Agent and the Banks may sustain or incur as a consequence of (a) default by Borrowers in payment when due of the principal amount of or interest on any Eurodollar Loans, (b) failure of Borrowers to make, or convert the Interest Rate of, a Eurodollar Loan after Borrowers have given (or are deemed to have given) an Interest Rate Request, and/or (c) the making by Borrowers of a payment or prepayment of a Eurodollar Loan on a day which is not the last day of the Interest Period with respect thereto, calculated as provided in the following paragraph. This agreement and covenant shall survive terminations or expiration of this Agreement and payment of the Obligations.

Contemporaneously with any payment or prepayment of principal of a Eurodollar Loan, a prepayment fee shall be due and payable to the Agent and each Bank in an amount equal to the product of

(A) the amount so prepaid the Agent or to such Bank

multiplied by

- (B) the difference (but not less than zero) of
  - (i) the constant maturity 360-day interest yield (as of the first day of the then effective Interest Period and expressed as a decimal)

minus

- - ----

multiplied by

of the first day of the then effective Interest Period, a remaining term approximately equal to the original Interest Period, (ii) the 360-day interest yield (as of the Banking Day immediately

for a United States Treasury bill, note, or bond (a "Treasury Obligation") selected by the Agent, in an aggregate amount comparable to the amount prepaid, and having, as

the Banking Day immediately preceding the prepayment date and expressed as a decimal) on such Treasury obligation and having, as of the Banking Day immediately preceding the prepayment date, a remaining term until maturity approximately equal to the unexpired portion of the Interest Period,

## (C) the quotient of

(i)

the number of calendar days in the unexpired portion of the Interest Period, divided by

(ii) 360.

The applicable yields on the Treasury Obligations described above shall be determined based upon the Federal Reserve Statistical Release H.15 published for the applicable determination dates set forth above. Any Treasury Obligation selected when the related Interest Period is one (1) year or less shall be United States Treasury Bills. Neither the Agent nor the Banks shall be obligated or required to have actually reinvested the prepaid amount of the Eurodollar Loan in any such Treasury Obligation as a condition precedent to Borrowers' being obligated to pay a prepayment fee as outlined above. Neither the Agent nor the Banks shall be obligated to accept any prepayment of principal unless it is accompanied by the prepayment fee, if any due in connection therewith as calculated pursuant to the provisions of this paragraph. No prepayment fee payable in connection herewith shall in any event or under any circumstances be deemed or construed as a penalty.

#### 2.5 Fees.

(a) Commitment Fee. Borrowers shall pay each Bank a commitment fee (the "Commitment Fee") for the period commencing on the Closing Date up to and including the Maturity Date, equal to one-quarter of one percent (0.25%) per annum (computed on the basis of the actual number of days elapsed over a 365 or 366 (as the case may be) day year) of the average daily unused portion of such Bank's percentage of the Commitment, as reduced from time to time pursuant to Section 2.6 or 2.7. The Commitment Fee shall be payable to each Bank quarterly in arrears on the first Banking Day of January, April, July and October of each year and on the date of termination of the Commitment.

(b) Audit Fee. Borrowers shall pay to the Agent an audit fee (the "Audit Fee") in an initial amount of \$5,000 per quarter for the period commencing on the Closing Date up to and including the Maturity Date. The Audit Fee shall be subject to reasonable adjustment in the Agent's sole and absolute discretion and shall be payable in advance on the first Banking Day of January, April, July and October of each year. Borrowers shall also reimburse the Agent for direct expenses incurred by the Agent in connection with audit services performed by employees or agents of the Agent.

(c) Facility Fee. Borrowers shall pay to the Agent a facility fee of \$25,000 on the Closing Date.

(d) Administrative Fee. Borrowers shall pay to the Agent an administrative fee (the "Administrative Fee") in an amount equal to \$15,000 per quarter for the period commencing on the Closing Date up to and including the Maturity Date. The Administrative Fee shall be payable in advance on the first Banking Day of January, April, July and October of each year.

2.6 Early Termination Fee. In the event of the termination by, or on behalf of Borrowers, of the Commitment (unless that termination is accompanied by the repayment of the Loans from the proceeds of the sale of substantially all of the assets of any Borrower or Subsidiary or an initial public offering of the common stock of any Borrower of Subsidiary and not from the proceeds of loans or other debt), Borrowers shall pay a fee (the "Early Termination Fee") equal to the following amount:

\$675,000 \$450,000 \$225,000

Da	t	e
----	---	---

Closing Dat	e through	July 1,	1998	
July 2, 199	8 through	July 1,	1999	
On or after	July 2, :	1999		

In the event of the termination of the Commitment and repayment of the Loans from the proceeds of the sale of substantially all of the assets of one or more of Borrowers or the Subsidiaries or an initial public offering of the common stock of one or more of Borrowers or Subsidiaries, the Early Termination Fee shall be \$75,000.

Simultaneously with the termination of the Commitment pursuant to this Section 2.6, Borrowers shall (i) pay to each Bank the percentage share of the total Commitment Fee of such Bank's pro-rata portion of the Commitment accrued to the date of such termination and (ii) subject to the provisions of Sections 2.10 and 2.11 prepay each Bank's Note(s) without penalty or premium in an amount equal to the unpaid principal balance of each such Note outstanding plus interest accrued but unpaid thereon on such date.

2.7 Mandatory Prepayments; Mandatory Reductions of Commitment.

(a) If, at any time, the Loan Outstandings to any Bank plus interest accrued but unpaid thereon exceed such Bank's percentage share of the Commitment, Borrowers shall make mandatory prepayments to such Bank in such amounts as may be necessary to eliminate such excess.

(b) Simultaneously with any termination or reduction of the Commitment pursuant to this Section 2.7, Borrowers shall (i) pay to each Bank the Commitment Fee on the terminated or reduced portion of its percentage share of the Commitment accrued to the date of such termination or reduction, (ii) in the event of termination in full of the Commitment, pay the Early Termination Fee; and (iii) subject to the provisions of Sections 2.10 and 2.11, prepay each such Bank's Note(s) without penalty or premium in a principal amount equal to the amount, if any, by which the Loan Outstandings on such date exceed the Commitment on such date, after giving effect to such termination.

(c) At the option of the Agent, there shall be a mandatory prepayment of the Loans in an amount equal to 100% of the net cash proceeds received by any Borrower or any Subsidiary from the sale or issuance of its common stock in one or more public offerings pursuant to an effective registration statement under the Securities Act (other than a registration statement on Form S-8) after the date hereof.

(d) All prepayments referred to in this Section 2.7 shall be applied pro rata among the Banks.

(e) At the option of the Agent, there shall be a Commitment reduction in the sole discretion of the Banks of 100% of the net cash proceeds (after payment of related taxes and expenses) from the sale of assets (other than the Excepted Collateral) received by a Borrower or a Subsidiary that results in net cash proceeds in excess of \$250,000; provided, however, that entering into a sale-leaseback transaction shall not be considered a sale of assets for purposes of this provision.

(f) At the option of the Agent, there shall be a mandatory prepayment of the Loans in full in the event that any prepayment, or any event which causes a prepayment, of all or any portion of the outstanding principal amount of the Investor Notes or the enterWorks Notes occurs. Such events which may cause a prepayment of the Investor Notes or the enterWorks Notes may include but are not limited to: (i) the distribution and sale of common stock of a Borrower or Subsidiary pursuant to an effective registration statement (some of the proceeds of which sale are available to such Borrower or Subsidiary) in one or more public offerings (other than a registration statement on Form S-4 or Form S-8) which has been filed with the Commission and has become effective; (ii) any sale of securities of a Borrower or a Subsidiary, which results in net cash proceeds to such Borrower or a Subsidiary in excess of \$1,000,000, other than (A) obligations for borrowed money due within one year and (B) other obligations for money borrowed from the Banks and/or their successors, substitutes and participants and their respective assigns and any refinancing thereof; or (iii) a Merger. For purposes of this Section 2.7(f), "Merger" shall mean (x) a merger, consolidation or other combination (1) to which a Borrower or a Subsidiary is a party, but in which such Borrower or a Subsidiary is not the surviving corporation or (2) which results in the acquisition of "beneficial ownership" of securities of a Borrower or a Subsidiary that represent fifty percent (50%) or more of the total number of votes that may be cast for the election of directors by any "person" or "group" (as such terms are defined in Rule 13(d) promulgated under the Exchange Act), or (y) a sale by a Borrower or a Subsidiary of all or substantially all of its assets.

2.8 Form and Terms of Payment.

(a) All payments by Borrowers of principal of or interest on the Notes and of any fee due hereunder shall be made at the addresses of the Banks hereinafter set forth in this Agreement (or at such other address as any Bank shall have furnished to Borrowers in writing) and shall be made in United States Dollars in immediately available funds. All payments by Borrowers of any amount due to the Agent hereunder shall be made in U.S. Dollars in immediately available funds at the address of the Agent as hereinafter set forth (or at such other address as the Agent shall have furnished to Borrowers in writing).

(b) Whenever any payment to be made hereunder or under any Note shall become due on a day that is not a Banking Day, such payment shall be made on the next succeeding Banking Day. If the date for any payment of principal is extended by operation of the foregoing sentence or law or otherwise, interest thereon shall be payable for such extended period.

(c) Any prepayment of principal of any Loans shall be accompanied by accrued and unpaid interest on the principal so prepaid through the date of payment.

2.9 Late Payments. If Borrowers shall fail to make any payment of principal of or interest on the Note of any Bank or any other amount due hereunder when the same becomes due and payable, whether at maturity or at a date fixed for the payment of any installment or prepayment thereof or by declaration, acceleration or otherwise, Borrowers shall pay to such Bank, on demand, an additional amount as a premium on such unpaid amount from the date due until paid in full at a rate or rates per annum equal to two percent (2%) above the rate or rates otherwise applicable in accordance with the terms of this Agreement.

2.10 Increased Costs and Reduced Return.

(a) Increased Costs. In the event that any Legal Requirement or any change in a Legal Requirement or other law or regulation or in the interpretation or application thereof or compliance by any Bank with any request or directive (whether or not having the force of law) from any central bank or other governmental authority:

(i) does or shall subject such Bank to any tax of any kind whatsoever with respect to this Agreement or any Loan or change the basis of taxation ofpayments to such Bank of principal, commitment fees, interest or any other amount payable hereunder (except for changes in the rate of taxes on the overallnet income of such Bank);

(ii) does or shall impose, modify or hold applicable any reserve, special deposit, compulsory loan or similar requirement against assets held by, or deposits or other liabilities in or for the account of, advances or loans by, or other credit extended by, or any other acquisition of funds by, any office of such Bank; or

(iii) does or shall impose on such Bank any other condition; and the result of any of the foregoing is to increase the cost to such Bank of making, renewing or maintaining any Loan, any portion of the Commitment or to reduce any amount receivable hereunder, then, in any such case, Borrowers shall promptly pay such Bank, upon its demand, any additional amounts necessary to compensate such Bank for such additional cost actually incurred or reduced amount to be received which such Bank, in its sole discretion, deems to be material. If any Bank becomes entitled to claim any additional amounts pursuant to the foregoing sentence, it shall promptly notify Borrowers with a copy to the Agent of the event by reason of which it has become so entitled. A certificate as to any additional amounts payable pursuant to this Section 2.10(a) submitted by such Bank to Borrowers shall be conclusive in the absence of manifest error. This covenant shall survive the termination of this Agreement and repayment of the Loans. All such amounts shall be invoiced separately by a Bank and shall be paid by Borrowers within five (5) days of receipt by Borrowers of such invoice.

(b) Reduction of Rate of Return. In the event that any Bank shall have determined that the adoption of any law, rule or regulation regarding capital adequacy, or any change therein or in the interpretation or application thereof or compliance by such Bank or any corporation controlling such Bank with any request or directive regarding capital adequacy (whether or not having the force of law) from any central bank or governmental authority, does or shall have the effect of reducing the rate of return on such Bank's or such corporation's capital as a consequence of its obligations hereunder to a level below that which such Bank or such corporation could have achieved but for such adoption, change or compliance (taking into consideration such Bank's or such corporation's policies with respect to capital adequacy) by an amount deemed by such Bank to be material, then, from time to time, after submission by such Bank such additional amount or amounts as will compensate such Bank for the actual amount of such reduction. A certificate as to any additional amounts payable pursuant to this Section 2.10(b) submitted by a Bank to Borrowers shall be invoiced separately by a Bank and shall be paid by Borrowers within five (5) days of receipt by Borrowers of such invoice.

(c) Conversion of Loans. Notwithstanding Sections 2.10 (a) and (b) above, if any Bank incurs the increased costs described in Section 2.10(a) above or

makes the determination described in Section 2.10(b) above, then all Loans of such Bank then outstanding and affected thereby shall be converted into Loans which are not affected thereby (if not otherwise prohibited under the terms of this Agreement).

2.11 Funding Losses. Borrowers agree to indemnify each Bank and to hold each Bank harmless from any loss or expense which such Bank may sustain or incur as a consequence of (i) default by Borrowers in payment when due of the principal amount of or interest on any Loan or any other amount payable by Borrowers hereunder, (ii) default by Borrowers in making a borrowing after Borrowers have given a notice of borrowing or a notice of conversion and/or (iii) default by Borrowers in making any prepayment after Borrowers have given a notice. This covenant shall survive termination of this Agreement and repayment of the Loans. All such amounts shall be invoiced separately by a Bank and shall be paid by Borrowers within five (5) days after receipt by Borrowers of such invoice.

2.12 Taxes. Each Bank shall receive payments of amounts of principal of and interest on the Note of such Bank without deduction for any Taxes. If (i) any Bank shall be subject to any Tax in respect of any Loan or (ii) Borrowers shall be required to withhold or deduct any such Tax from any such amount, the interest rate otherwise payable hereunder shall be adjusted to reflect all additional costs incurred by such Bank in connection with the payments by such Bank or the withholding by Borrowers of such Tax and Borrowers shall provide such Bank with a statement detailing the amount of any such Tax actually paid by Borrowers. Determination by a Bank of the amount of such costs shall, in the absence of manifest error, be conclusive. All such amounts shall be invoiced receipt by Borrowers of such invoice.

2.13 Security. The Obligations shall be secured by and entitled to the benefits of (i) a first priority perfected security interest in the Collateral pursuant to the terms of the Security Agreements and the Pledge Agreements; (ii) an assignment of monies due or to become due meeting all of the requirements of the Assignment of Claims Act for all Government Contracts (provided, however, that with respect to monies due and to become due in relation to Government Contracts, the ultimate funded value of which will not exceed \$1,000,000, the Agent may, in its sole discretion, not file the "Notice of Assignment" and other documents required by the Assignment of Claims Act); (iii) the Assignment of Tenant's Interest under Leases; and (iv) the Guaranties of Subsidiaries.

2.14 Loan Account. The Agent will establish and maintain an account on its books (the "Loan Account") to which (a) the principal amount of each advance of the Loan made by the Agent shall be debited on the date made, (b) each payment made by Borrowers to the Agent for credit to the Loan Account shall be credited on the date received, (c) all or any part of the collected funds (collectively, the "Collections") credited to the Collateral Account which are to be credited to the Loan Account shall be credited on the date such credit is made in the amount of each such Collection, (d) all accrued interest on the Loan and any Enforcement Costs shall be debited on the date such interest or Enforcement Costs became past due, (e) all Enforcement Costs shall be debited on the date the same become due and (f) all fees due to the Banks and the Agent shall be debited by recording therein on the date imposed the amount of such charges or fees. All credit entries to the Loan Account are conditional and shall be readjusted as of the date made if final payment is not received by the Agent in cash or solvent credits. The Agent shall not less frequently than monthly render a statement or statements of account to Borrowers with respect to the Loan Account.

2.15 Collateral Account. The Borrower will deposit or cause to be deposited to a bank account designated by the Agent and from which the Agent alone has power of access and withdrawal (the "Collateral Account") all Items of Payment. Borrowers shall deposit Items of Payment for credit to the Collateral Account on the day of the receipt thereof, and in precisely the form received, except for the endorsement of Borrowers where necessary to permit the collection of Items of Payment, which endorsement Borrowers hereby agrees to make. Pending such deposit, Borrowers will not commingle any Items of Payment with any of its other funds or property, but will hold them separate and apart. The Agent shall credit all of the Collections that exceed the aggregate debit balance of the Loan Account to a banking account of Borrowers with the Agent on the first Banking Day following the deposit of the Collections to the Collateral Account.

2.16 Illegality. Notwithstanding any other provisions herein, if any laws or regulations or any change therein or in the interpretation or application thereof shall make it unlawful for any Bank to make or maintain Eurodollar Loans as contemplated by this Agreement, (i) the commitment of such Bank hereunder to make Eurodollar Loans shall forthwith be canceled, and (ii) the Loans then outstanding as Eurodollar Loans, if any, shall be repaid and made as Base Rate Loans (if not otherwise prohibited under the terms of this Agreement) at the option of Borrowers in accordance with the election procedures set forth in Section 2.4(i); provided, however, that prior to the effective date of such election, interest shall be calculated at the Base Interest Rate. Any remaining commitment of such Bank hereunder to make Eurodollar Loans (but not other Loans) shall terminate forthwith and borrowings from such Bank at a time when borrowings from the other banks are to be of Eurodollar Loans shall be by way of Base Rate Loans as provided herein. Upon the occurrence of any such change, such Bank shall promptly notify Borrowers thereof, and shall furnish to Borrowers in writing evidence thereof certified by such Bank.

If any repayment to such Bank of any Eurodollar Loan (including conversions thereof) is made under this Section 2.16 on a day other than a day otherwise scheduled for a payment of principal of or interest on such Loan, Borrowers jointly and severally shall pay to such Bank upon its request such amount or amounts as will compensate it for the amount by which the rate of interest on such Loan immediately prior to such repayment exceeds the stated rate of interest on relending or reinvesting the funds received in connection with such prepayment, in each case for the period from the date of such prepayment to the Banking Day next succeeding the last day of such then current Interest Period, all as determined by such Bank in its good faith discretion.

#### ARTICLE III

#### LETTERS OF CREDIT

3.1 Issuance. Borrowers and the Banks acknowledge that from time to time Borrowers may request that a Bank issue a standby letter or letters of credit (each being herein referred to as a "Letter of Credit"), in connection with Borrowers' business operations. Any such request by Borrowers shall be made by submitting to such Bank an Application and Agreement for Standby Letter of Credit (each being herein referred to as an "Application") on such Bank's standard form, which Application shall be executed by authorized officers of each Borrower, and be accompanied by such other supporting documentation and information, borrowing and other corporate resolutions and opinions of counsel as such Bank may from time to time request. Each Application shall be deemed to govern the terms of issuance of the subject Letter of Credit; it being understood, however, that the fees for issuance thereof shall be as set forth in Section 3.2.

3.2 Fees. In addition to the payment of all fees as set forth in Section 2.5 and Article 13, Borrowers shall pay to the issuing Bank the following for each Letter of Credit issued by such Bank for the benefit of Borrowers: (i) a one-time origination fee in the amount of \$300.00, payable upon the issuance of each such Letter of Credit, and (ii) for each three (3) month period (or partial three (3) month period) during which such Letter of Credit remains outstanding, a fee (the "Periodic LC Fee") in the amount of one-half of one percent (0.50%) of the original issued amount of such Letter of Credit, with the first Periodic LC Fee payment being payable on the date of issuance of such Letter of Credit, and each subsequent Periodic LC Fee payment being payable in advance on the first Banking Day of each and every successive third (3rd) calendar month after the Issuance Effective Date until such Letter of Credit is canceled or expires. For purposes of this Section 3.2, the "Issuance Effective Date" shall be deemed to be the first day of the calendar month in which a Letter of Credit is issued (irrespective of the actual date of issuance). Notwithstanding the foregoing, it is expressly understood and agreed that the Banks shall not issue any Letters of Credit having an expiration date beyond the Maturity Date.

3.3 Other Matters. In the event a Bank issues any Letter(s) of Credit, (i) the maximum amount of credit available under the Commitment shall be reduced by the issued amount of each Letter of Credit (so long as such Letter of Credit remains outstanding), and (ii) any amounts drawn under any Letter of Credit shall be deemed a Loan and shall be deemed to be an advance under the Note, shall bear interest and be payable in accordance with the terms of the Note and shall be secured by the Security Documents (in the same manner as all other sums advanced under the Note). It is expressly understood and agreed that all obligations and liabilities of Borrowers to such Bank in connection with any such Letter(s) of Credit shall be deemed to be Obligations, and such Bank shall have no obligation to release its rights under the Security Documents until the Note and all other sums due to such Bank in connection with the Loans have been paid and satisfied in full, all Letters of Credit have been canceled or expired (i.e., none are outstanding), and such Bank has no further obligation or responsibility to make additional Loans or issue additional Letters of Credit. Furthermore, in no event whatsoever shall such Bank have any obligation to issue any Letter of Credit which would cause the issued amount of all then outstanding Letters of Credit issued by such Bank for the benefit of Borrowers to exceed, in the aggregate, \$300,000.

# ARTICLE IV

## REPRESENTATIONS AND WARRANTIES

In order to induce the Banks to enter into this Agreement and to make the Loans as provided hereunder, Borrowers make the representations and warranties set forth in Section 4.1 through 4.35 inclusive as of the Closing Date, on the date of each new Loan and as of any date on which such representation and/or warranty is stated to be made herein.

4.1 Organization, Standing, etc. of Telos Corp. Telos Corp. is a corporation duly organized, validly existing and in good standing under the laws of the State of Maryland. Telos Corp. is qualified to do business as a foreign corporation in each jurisdiction in which the character of its business or the nature or location of the assets it owns or leases would require Telos Corp. to be qualified and where failure so to qualify would have a material adverse effect on its business, assets, operations or conditions, financial or otherwise. Telos Corp. has all requisite corporate power and authority to own and operate its properties and to carry on its business as now conducted and ascontemplated to be conducted. Telos Corp. does not own any stock or other equity interest in any Person other than the entities listed on Schedule 4.1.

4.2 Organization, Standing, etc. of Telos Sub and the Subsidiaries. Telos Sub and each of the Subsidiaries are corporations duly organized, validly existing and in good standing under the laws of their respective states of incorporation. Telos Sub and each of the Subsidiaries are qualified to do business as foreign corporations in each jurisdiction in which the character of their respective businesses or the nature or location of the assets they own or lease would require them to be so qualified and where failure so to qualify would have a material adverse effect on their respective businesses, assets, operations or conditions, financial or otherwise. Telos Sub and each of the Subsidiaries have all requisite corporate power and authority to own and operate their respective properties and to carry on their respective businesses as now conducted and as contemplated to be conducted. Neither Telos Sub nor any of the Subsidiaries owns any stock or other equity interest in any Person other than the entities listed on Schedule 4.2.

4.3 Corporate Authority. Each of Telos Corp., Telos Sub and each of the Subsidiaries has the full corporate power and authority to execute, deliver and perform the Revolving Loan Documents to which it is a party. Each of Telos Corp., Telos Sub and each of the Subsidiaries has taken all action required by law and its articles of incorporation and bylaws to authorize and approve the execution, delivery and performance of the Revolving Loan Documents to which it is party and the transactions contemplated thereby.

4.4 Valid and Binding Obligations. The Revolving Loan Documents have been duly executed and delivered by each of Telos Corp., Telos Sub and each of the Subsidiaries party thereto and are valid and binding obligations of each of Telos Corp., Telos Sub and each of the Subsidiaries party thereto, enforceable against each of Telos Corp., Telos Sub and each of the Subsidiaries party thereto in accordance with their respective terms, except as enforceability may be limited by applicable bankruptcy, moratorium and similar laws of general application affecting the rights of creditors generally, and except as such enforceability may be limited by applicable principles of equity, whether applied by a court of equity or a court of law.

4.5 Authorized and Outstanding Capital Stock; Subsidiaries. Schedule 4.5 attached hereto correctly sets forth, as to each of Telos Corp., Telos Sub and each of the Subsidiaries, the number of authorized shares, the number of shares of each class of capital stock outstanding and the identity of the owners of such outstanding shares. All of the outstanding capital stock of each of Telos Corp., Telos Sub and each of the Subsidiaries is validly issued, fully paid and nonassessable and is owned by the Persons identified on said Schedule, free and clear of all Liens. All of the Pledged Stock and the enterWorks Stock listed on Schedule 4.5 has been validly issued, is fully paid and nonassessable and is owned free and clear of all Liens, except for Liens created by the Pledge Agreements. Except as stated in Schedule 4.5, neither Telos Corp. nor Telos Sub nor any Subsidiary has granted or issued, or agreed to grant or issue, any options, warrants or similar rights to acquire or receive any of the authorized but unissued shares of its capital stock.

4.6 Compliance with Laws; Non-Contravention. Neither any Borrower nor any Subsidiary is bound by any agreement or instrument or subject to any charter or other corporate restriction which materially and adversely affects its business, assets, operations or condition, financial or otherwise. Neither any Borrower nor any Subsidiary is in violation of its Articles of Incorporation or By-Laws or of any statute, law, rule or regulation, or of any judgment, decree, writ, injunction, order or award of any arbitrator, court or governmental authority applicable to it, other than violations of statutes, laws, rules or regulations which do not have a material adverse effect on the business, financial condition or operations of a Borrower or a Subsidiary or on the transactions contemplated by the Revolving Loan Documents. Neither the authorization, execution and delivery of the Revolving Loan Documents, the consummation of the transactions herein or therein contemplated nor the fulfillment of or compliance with the terms of the Revolving Loan Documents, conflicts with or results or will conflict with or will result, in a breach of any of the terms of the Articles of Incorporation or By-Laws of a Borrower or a Subsidiary or any other corporate restriction or of any material statute, law, rule or regulation or of any judgment, decree, writ, injunction, order or decree of any arbitrator, court or governmental authority, or of any material agreement or instrument, which is applicable to a Borrower or a Subsidiary or by which a Borrower or a Subsidiary or any assets thereof may be subject or bound, or constitutes or will constitute a default thereunder, or results or will result in the imposition of any material Lien upon any of the properties or assets of a Borrower or a Subsidiary, except for Permitted Liens.

4.7 Financial Information, etc. All of the financial statements and other reports provided by Borrowers to the Agent (the "Financial Statements") (i) are complete and correct in all respects, (ii) have been prepared in accordance with GAAP and (iii) fairly present the financial position and results of operations, as the case may be, of Borrowers and the Subsidiaries as of the dates and for the periods indicated, subject to normal year-end adjustments that will not, in the aggregate, be material. There are no material liabilities of a Borrower or a Subsidiary, direct or indirect, absolute or contingent, as of the dates of the Financial Statements that are not reflected therein or in the notes thereto.

4.8 No Material Adverse Change. Since December 31, 1996, there has been no material adverse change in the businesses, properties, prospects, operations, liabilities or condition (financial or otherwise) of any of Telos Corp., Telos Sub or the Subsidiaries, individually or on a consolidated basis, except as otherwise expressly disclosed to the Agent in writing not later than forty-eight (48) hours prior to the Closing Date.

4.9 The Collateral. Except for the security interests granted in the Security Documents, each of the Borrowers and the Subsidiaries is, or, to the extent that the Collateral will be acquired after the date hereof, will be, the legal and beneficial owner of the Collateral, free and clear of all Liens, except Permitted Liens. Except for the Security Documents, neither any Borrower nor any Subsidiary has executed any other security or similar agreement with respect to the Collateral or the Pledged Stock and neither any Borrower nor any Subsidiary has heretofore given any Lien or security interest of any kind (whether purchase money or otherwise) in the Collateral or the Pledged Stock to any Person. The security interest granted by the Security Documents in the Collateral and the Pledged Stock shall constitute a first priority security interest and Lien in the Collateral and the Pledged Stock as provided by the Uniform Commercial Code in effect in the appropriate jurisdictions. Neither any Borrower or any Subsidiary has executed any security agreement, pledge agreement or similar agreement with respect to the enterWorks Stock and neither any Borrower or any Subsidiary has given any Lien or security interest of any kind (whether purchase money or otherwise) in the enterWorks Stock to any Person. All information with respect to the Collateral, the Pledged Stock and the enterWorks Stock set forth in any schedule furnished at any time by a Borrower to the Agent hereunder is and will be true and complete in all material respects as of the date furnished.

4.10 Eligible Receivables. All Eligible Receivables are and shall be good and valid accounts representing undisputed bona fide indebtedness of the account debtor to a Borrower or a Subsidiary, as the case may be, for merchandise theretofore shipped or delivered pursuant to or in connection with a contract for sale, or for services theretofore performed by such Borrower or such Subsidiary to or for said account debtor. There are and shall be no setoffs or counterclaims of any nature whatsoever against any Eligible Receivable. No agreement under which any deduction or discount may be claimed has been or will be made with the account debtor of any of the Eligible Receivables. Each of the Borrowers and the Subsidiaries is and shall be the lawful owner of all Eligible Receivables and has and shall have the unrestricted right to pledge, sell, assign and transfer the Eligible Receivables and grant the security interest granted herein and in the Security Documents. This representation shall be deemed made as of the date Borrowers report to the Agent that an Eligible Receivable exists, including, without limitation, the date of each Borrowing Base Certificate and as of the date a certificate, report or statement concerning Eligible Receivables is delivered by or on behalf of Borrowers pursuant to this Agreement.

4.11 Licenses; Franchises; etc. Schedule 4.11 attached hereto accurately and completely lists all authorizations, licenses, permits and franchises of any public or governmental regulatory body (i) granted or assigned to a Borrower or a Subsidiary and (ii) which are necessary for the conduct of the business of a Borrower or a Subsidiary, as now conducted and proposed to be conducted except (as to authorizations, licenses, permits and franchises other than those required for the performance of U.S. Government Contracts) where the failure to have fulfilled or performed its obligations thereunder does not materially and adversely affect the business, financial condition or operations of a Borrower or a Subsidiary (such authorizations, licenses, permits and franchises, together with all extensions or renewals thereof, being herein sometimes referred to, collectively, as the "Franchises"). All of such Franchises are validly issued and in full force and effect and each Borrower and each Subsidiary has fulfilled and performed all of its obligations with respect thereto required to have been fulfilled and performed and has full power and authority to operate thereunder, except (as to Franchises other than those required for the performance of U.S. Government Contracts) where failure so to have fulfilled or performed obligations does not materially and adversely affect the business, assets, operations or condition, financial or otherwise, of a Borrower or a Subsidiary.

4.12 Material Agreements. Schedule 4.12 attached hereto accurately and completely lists all material leases and all material agreements involving future payment or receipt of more than \$100,000 which are presently in effect ("Material Agreements") in connection with the conduct of the businesses of Borrowers and the Subsidiaries. All such leases and agreements are in full force and effect, and each of the Borrowers and the Subsidiaries has fulfilled and performed all of its material obligations with respect thereto required to have been fulfilled and performed. All Material Agreements in connection with the business of Borrowers and the Subsidiaries are in the name of Telos Corp. or Telos Sub. except as described on Schedule 4.12.

4.13 Tax Returns and Payments. Each of the Borrowers and the Subsidiaries has filed all tax returns required by law to be filed and has paid all Taxes, assessments and other governmental charges levied upon any of its properties, assets, income or franchises, other than those not yet delinquent, except payments for Taxes that have been deferred by such Borrower or such Subsidiary because such Borrower or such Subsidiary is contesting the imposition or amount of such Taxes in good faith and has made adequate reserves on its books therefor. The charges, accruals and reserves on the books of each such Borrower or Subsidiary, as the case may be, in respect of its Taxes are adequate, and neither any Borrower nor any Subsidiary knows of any actual or proposed assessment for additional Taxes for any fiscal period or of any basis therefor.

4.14 Indebtedness. Neither any Borrower nor any Subsidiary is in default beyond any grace period under any outstanding Indebtedness. Schedule 4.14 attached hereto correctly describes (i) all outstanding Indebtedness of the Borrowers and the Subsidiaries and (ii) all existing Liens in respect of any property or assets of Borrowers and the Subsidiaries. For purposes of this Section 4.14, the term "Indebtedness" shall not include obligations by a Borrower or a Subsidiary to vendors and suppliers incurred in the ordinary course of business.

4.15 Title to Properties. Borrowers and the Subsidiaries have good and marketable title to all of their respective properties and assets, and none of such properties or assets is subject to any Lien, except Permitted Liens. Except as set forth on Schedule 4.15, Borrowers and the Subsidiaries enjoy quiet possession under all leases to which they are a party as lessee, and all of such leases are valid, subsisting and in full force and effect. None of such leases contains any provision restricting the incurrence of indebtedness by the lessee or any unusual or burdensome provision materially adversely affecting the business, assets, operations or condition, financial or otherwise, of any Borrower or any Subsidiary.

4.16 Litigation. There are no actions, suits or proceedings at law or in equity or by or before any governmental instrumentality or other agency or regulatory authority now pending or, to Borrowers' best knowledge, threatened against or affecting any Borrower or any Subsidiary which, if adversely determined, could, individually or in the aggregate, (i) materially impair the ability of such Borrower or such Subsidiary to conduct its business as now conducted, (ii) materially and adversely affect the business, assets, operations or condition, financial or otherwise, of such Borrower or such Subsidiary or (iii) impair the validity or enforceability of, or the ability of such Borrower or such Subsidiary to perform its obligations under, any of the Revolving Loan Documents.

4.17 Approvals; Consents. Neither any Borrower nor any Subsidiary nor any of their respective shareholders is required to obtain any order, consent, approval or authorization of, or required to make any declaration or filing with, any governmental authority in connection with (i) the execution and delivery by Borrowers or the Subsidiaries of the Revolving Loan Documents, (ii) the granting by any Borrower or any Subsidiary of the security interests in the Collateral and the Pledged Stock pursuant to the Security Documents or (iii) the ability of any Borrower or any Subsidiary to conduct its business in the same manner after the date hereof as such business has been conducted prior to the date hereof (other than ministerial duties to be performed in connection with any such order, consent, approval or authorization). Schedule 4.17 sets forth all orders, consents, approvals and authorizations of governmental authorities which are necessary to continue the businesses of Borrowers and the Subsidiaries (collectively, the "Consents"). All material Consents have been obtained and are in full force and effect.

## 4.18 Compliance with ERISA.

(a) The consummation of any of the transactions contemplated by any of the Revolving Loan Documents is not and will not constitute a "prohibited transaction" within the meaning of Section 4975 of the Code, or Section 406 of ERISA. None of the Plans is a multi-employer plan, as defined in Section 4001(a)(3) of ERISA (a "Multi-employer Plan") and no Multi-employer Plan has been maintained or contributed to at any time during the five calendar years preceding this Agreement. There has been no material "prohibited transaction," as such term is defined in Section 406 of ERISA and Section 4975 of the Code, with respect to any Plan, which could result in any material liability of any Borrower or any Subsidiary. All Plans are, to the knowledge of Borrowers, in compliance in all material respects with the requirements prescribed by any andall statutes (including ERISA and the Code), orders, or governmental rules and regulations currently in effect with respect thereto. Each Plan intended to qualify under Section 401(a) of the Code and each trust intended to qualify under Section 501(a) of the Code does so qualify. Neither any Borrower nor any Subsidiary maintains any defined benefit plan, as such term is defined in Section 414(j) of the Code.

by a Borrower or a Subsidiary, as appropriate or by an ERISA Affiliate, as applicable, of all amounts which are required under the terms of each Plan to have been paid as contributions to such Plan. There has been no termination of a Plan or trust created under any Plan that would give rise to a material liability to the PBGC on the part of a Borrower or a Subsidiary, as appropriate, or an ERISA Affiliate. No material liability to the PBGC has been or is expected to be incurred with respect to any Plan by a Borrower or a Subsidiary, as appropriate, or an ERISA Affiliate. The PBGC has not instituted proceedings to terminate any Plan which is maintained or is to be maintained by a Borrower or a Subsidiary, as appropriate, or an ERISA Affiliate. There exists no condition or set of circumstances which present a material risk of termination or partial termination of any Plan by the PBGC. No Plan has an "accumulated funding deficiency" (whether or not waived) within the meaning of Section 412 of the Code or Section 302 of ERISA.

(b) No event has occurred that could give rise to any lien, constructive or otherwise and no lien, constructive or otherwise, has arisen under ss.ss. 4068(a) or 302(f) of ERISA with respect to any defined benefit pension plan that could affect any asset of a Borrower or a Subsidiary.

(c) Neither any Borrower nor any Subsidiary has maintained, or maintains, any plan, fund, program, commitment, or arrangement pursuant to which it provides, or promises to provide, to current or former employees postretirement benefits other than pensions within the meaning of Statement of Financial Accounting Standard No. 106.

4.19 Brokers, etc. Neither any Borrower nor any Subsidiary has taken any action which would obligate any holder of the Notes to pay a broker's fee, finder's fee or commission in connection with the transactions contemplated by the Revolving Loan Documents.

4.20 Compliance with Environmental Laws. Each of the Borrowers and the Subsidiaries is in compliance with all applicable statutes, laws, rules, regulations and orders of all governmental authorities relating to environmental protection and pollution control with respect to the conduct of its business and the ownership of its properties, except violations which do not in the aggregate have a material adverse effect on the business, financial condition or operations of such Borrower or such Subsidiary or on the transactions contemplated by this Agreement.

4.21 Status under Certain Statutes. Neither any Borrower nor any Subsidiary is (i) a "public utility company" or a "holding company," or an "affiliate" or a "subsidiary company" of a "holding company," or an "affiliate" of such a "subsidiary company," as such terms are defined in the Public Utility Holding Company Act of 1935, as amended, or (ii) a "public utility" as defined in the Federal Power Act, as amended, or (iii) an "investment company" or an "affiliated person" thereof or an "affiliate person" of any such "affiliated person" as such terms are defined in the Investment Company Act of 1940, as amended.

4.22 Disclosure. None of the Revolving Loan Documents or any other document, certificate, or statement furnished to any Bank or the Agent by or on behalf of a Borrower or a Subsidiary in connection with the transactions contemplated hereby contains any untrue statement of a material fact or omits to state a material fact necessary in order to make the statements contained herein and therein not misleading; provided, however, that (i) no statement that has been modified or superseded by a subsequent statement in writing shall be considered untrue or misleading, and (ii) financial projections shall not be considered untrue or misleading provided they were prepared based upon assumptions that were reasonable at the time the financial projections were made. There is no fact known to Borrowers which materially and adversely affects the business, properties, operations or condition, financial or otherwise, of a Borrower or a Subsidiary which has not been set forth in this Agreement, any of the other Revolving Loan Documents or otherwise disclosed in writing to the Banks.

4.23 Exchange Act Registration. Except for the Public Preferred Stock, no class of any security of any Borrower or any Subsidiary is now or is required to be registered with the Commission pursuant to Section 12 of the Exchange Act.

4.24 Use of Proceeds. The proceeds of the Loans will be used for the purpose of (i) working capital and (ii) Capital Expenditures, as permitted herein. Borrowers will not use any part of such proceeds for any other purpose, including, without limitation, any purpose which would involve a violation of Section 7 of the Exchange Act or Regulation G, T, U or X.

4.25 Suspension or Debarment. Neither any Borrower nor any Subsidiary nor any of their respective directors, officers or employees has received any notice of or information concerning any proposed or contemplated suspension or debarment from U.S. Government contracting. No basis exists for the default termination of any current U.S. Government Contract.

4.26 Claims and Investigations. There are no pending or threatened claims, investigations (whether formal or informal), litigation, disputes, protests, or other controversies involving any Borrower or any Subsidiary

pertaining to or arising out of any U.S. Government Contract which, if adversely determined, would have a material adverse effect on the business, assets, operations or condition, financial or otherwise, of any Borrower or any Subsidiary. Neither any Borrower nor any Subsidiary has filed or contemplates the filing of any claims or demands for payment against the U.S. Government or any other party arising out of or in connection with any U.S. Government Contract, other than progress billings, public vouchers, and invoices submitted in the ordinary course of business.

4.27 Violation of Contracting Statutes. Neither any Borrower nor any Subsidiary has committed any violation (which violation could individually or in the aggregate have a material adverse effect on the business, assets, operations or condition, financial or otherwise, of such Borrower or such Subsidiary) of any federal statute or regulation pertaining to or affecting U.S. Government contractors or U.S. Government procurement in general, including, without limitation, the following: the Armed Services Procurement Act of 1947; the Federal Property and Administrative Services Act of 1949; the False Claims Act; the False Statements Act; the U.S. espionage laws; the U.S. Export Administration Act; the U.S. Anti-Trust Statutes; the Anti-Kickback Act of 1986; laws pertaining to restrictions on private sector employment of former Government officers or employees (e.g., 10 U.S.C. ss.ss. 2397b and 2397c); laws prohibiting the payment of bribes, gratuities or contingent fees to Government personnel; the Competition in Contracting Act; except as otherwise provided in this Agreement and in the Security Agreements, the Assignment of Claims Act; the Anti-Deficiency Act; laws pertaining to the examination of books and records by the Comptroller General and other agencies of the Federal Government (e.g., 10 U.S.C. ss. 2313, 2306a); the Truth-In-Negotiations Act; laws pertaining to rights in technical data and computer software (e.g., 10 U.S.C. ss.ss. rights in technical data and computer software (e.g., 10 U.S.C. ss.ss. 2320-2321); laws pertaining to commercial pricing for spare or repair parts (e.g., 10 U.S.C. ss. 2323); laws pertaining to the allowability of costs under U.S. Government Contracts (e.g., 10 U.S.C. ss. 2324); laws pertaining to the prohibition against unreasonable limitations on subcontractors sales to the United States (e.g., 10 U.S.C. ss. 2402); labor laws pertaining to equal employment opportunity and affirmative action, minimum wages, and other labor standards; the Walsh-Healey Public Contracts Act; laws pertaining to employment of and affirmative actions for the handicapped and Vietnam era veterans; the Small Business Act; the Buy American Act; DOD Appropriation Acts; the Federal Acquisition Regulations; the Department of Defense FAR Supplement; regulations pertaining to Cost Accounting Standards; and all other statutes and regulations pertaining in any way to U.S. Government Contracting.

4.28 U.S. Government Contracts. No U.S. Government Contract of either Borrower or any Subsidiary contains a clause prohibiting or otherwise restricting the assignment of payments thereof or imposing any requirements with respect to any such assignment other than those provided in the Assignment of Claims Act generally. All U.S. Government Contracts of Borrowers and the Subsidiaries are in full force and effect. None of the transactions that have occurred or are contemplated to occur under the Revolving Loan Documents has or will result in (i) any material default or any other government action precluding any Borrower or any Subsidiary from performing any U.S. Government Contract, (ii) any withdrawal or termination of any Consent or material Franchise necessary for the performance by any Borrower or any Subsidiary of any U.S. Government Contract or (iii) violation of any federal statute or regulation pertaining to or affecting U.S. Government contractors or U.S. Government procurement. All U.S. Government Contracts of Borrowers and the Subsidiaries are in good standing, full force and effect, and all material obligations thereunder have been performed in accordance with their respective terms by Borrowers and the Subsidiaries. Schedule 4.28 attached hereto accurately and completely lists all U.S. Government Contracts of the New Subsidiaries with a monetary value of \$250,000 or more.

4.29 Subcontractors. Neither Borrower nor any Subsidiary has knowledge that any supplier or subcontractor to any Borrower or any Subsidiary in connection with a U.S. Government Contract is in material default or is unable to fulfill any material obligation to such Borrower or to such Subsidiary in connection with any such U.S. Government Contract.

4.30 No Assignment of Payments. No payments due or to become due under any U.S. Government Contract held by any Borrower or any Subsidiary have been assigned to any third party except as provided for in this Agreement.

4.31 Audits. Neither any Borrower nor any Subsidiary is undergoing any audit (other than normal and ordinary audits of a routine nature) or to the best knowledge of Borrowers, any investigation by the Defense Contract Audit Agency, the Inspector General of any federal agency, or any other governmental entity in connection with any U.S. Government Contract.

4.32 Security Clearances. Each of the Borrowers and the Subsidiaries has in full force and effect all necessary security clearances, including facility clearances and personnel clearances necessary to perform all of its contractual obligations. No U.S. Government Contract held by a Borrower or a Subsidiary involves the services of any employee or consultant who is not a U.S. citizen or an immigrant alien within the meaning of the U.S. Industrial Security Regulation and U.S. Industrial Security Manual, other than as approved by the appropriate government agency. No U.S. Government Contract held by a Borrower or a Subsidiary involves orrequires access to classified information other than "Secret," "Top Secret" or "Confidential."

4.33 Certain Agreements. Neither Borrower nor any Subsidiary is a party to or bound by any Teaming Agreement or similar "Contractor Team Arrangement" within the meaning of Federal Acquisition Regulations, Subpart 9.6, which (i) restricts the right of a Borrower or a Subsidiary to compete independently or with others for any procurement opportunity other than the specific U.S. Government contracting program enumerated in such Teaming Agreement, or (ii) contains any provision obligating a Borrower or a Subsidiary to pay any costs or expenses of any other party to such Teaming Agreement, other than payments for non-recurring engineering costs incurred in the ordinary course of business.

4.34 Export Control Laws. Each of the Borrowers and the Subsidiaries and all of their respective officers, members of the Board of Directors, employees, agents and representatives have operated in the past and are operating as of the Closing Date in material compliance with all applicable government export control laws and regulations including, but not limited to, the provisions of the following U.S. Government Regulations: the Export Administration Regulations, the International Traffic in Arms Regulations, the Foreign Assets Control Regulations, the Cuban Assets Control Regulations, and the Libyan Sanctions Regulations, the material breach of which would have a material adverse effect on the business, assets, operations or condition, financial or otherwise, of a Borrower or a Subsidiary.

# ARTICLE V

## CONDITIONS OF EFFECTIVENESS OF THIS AGREEMENT AND LENDING

5.1 Effectiveness of this Agreement. The effectiveness of this Agreement is subject to satisfaction of the following conditions precedent on or prior to the Closing Date (unless otherwise specified), documentation of which shall be in form and substance satisfactory to counsel for the Banks:

(a) Absence of Litigation. No action, proceeding, regulation or legislation shall have been instituted, threatened or proposed before any court, governmental agency or legislative body to enjoin, restrain or prohibit, or to obtain damages in respect of, or which is related to the transactions contemplated in the Revolving Loan Documents which, in the reasonable judgment of the Agent, would likely cause Borrowers or the Subsidiaries to become unable to comply with the terms and conditions set forth herein, or in the other Revolving Loan Documents.

(b) Absence of Material Adverse Changes. As of the Closing Date, there shall have been no material adverse change in the business, properties, operations or condition (financial or otherwise) of Borrowers and the Subsidiaries on a consolidated basis since December 31, 1996, except as otherwise expressly disclosed to the Agent in writing not later than forty-eight (48) hours prior to the Closing Date.

(c) The Revolving Loan Documents. Each of the Borrowers and the Subsidiaries shall have executed and delivered each of the Revolving Loan Documents to which each is a party. Without limiting the generality of the foregoing, each of the New Subsidiaries shall have executed and delivered (i) a Guarantee of Subsidiary, substantially in the form attached as Exhibit B hereto, and (ii) a Security Agreement substantially in the form attached as Exhibit C hereto.

(d) Opinions and Certificates. The Agent shall have received:

(i) The favorable opinions of Messrs. McGuire Woods Battle & Booth L.L.P. counsel for Borrowers, dated as of the Closing Date, substantially in the form attached as Exhibit G hereto; and

(ii) A certificate, dated as of such date, signed on behalf of Borrowers by any one of their President, Chief Executive Officer, Chief Financial Officer or Vice President (each sometimes referred to hereinafter as an "Executive Officer"), certifying that the conditions specified in this Section 5.1 have been fulfilled and the representations and warranties set forth in Article IV hereto are true and correct.

(e) Corporate Documents. The Agent shall have received:

(i) Articles of Incorporation. Certified copies of each New Subsidiary's Articles of Incorporation and Bylaws;

(ii) Good Standing Certificates. Long-form good standing certificates from the State of incorporation of each Borrower and each Subsidiary dated not more than seven (7) days before the Closing Date; (iii) Resolutions. Certified copies of resolutions adopted by each Borrower's and each Subsidiary's board of directors in form and substance satisfactory to the Agent, authorizing the execution, delivery and performance of the Revolving Loan Documents;

(iv) Incumbency Certificates. Certificates of incumbency of each Borrower and each Subsidiary showing the signatures and corporate authority of the officers executing the Revolving Loan Documents; and

(v) Other Information. All other information and documents which the Banks or their respective counsel may reasonably request in connection with the transactions contemplated by the Revolving Loan Documents, such information and documents where appropriate to be certified by an Executive Officer or by the appropriate governmental authorities.

(f) Payment of Fees. The fees, or portion of fees, described in Sections 2.5 and 3.2 and Article 13, which are due as of the Closing Date shall have been paid, or contemporaneously will be paid on the Closing Date, in full by Borrowers.

(g) Legal Requirements. The transactions contemplated by the Revolving Loan Documents shall not violate any Legal Requirement, including, without limitation, Regulations G, T, U and X.

(h) Legal and Organizational Structure. All terms and conditions of the capital management, legal and organizational structure of Borrowers and the Subsidiaries shall be acceptable to the Agent in its sole and absolute discretion.

(i) U.C.C., Tax Lien and Judgment Searches. The Agent shall have received such title reports and lien and judgment searches as are necessary to demonstrate that title to the Collateral and Pledged Stock of Borrowers and the Subsidiaries is free and clear of all Liens except Permitted Liens, that all U.C.C.-1 financing statements required to perfect the security interests of the Agent have been duly filed in the appropriate offices, and that all pledges of uncertificated securities included in the Collateral have been duly registered so that, when the Loans are advanced, the security interests of the Agent and the Banks in the Collateral and the Pledged Stock will be valid, perfected and of first priority.

5.2 Each Loan. The obligation of each of the Banks to make a Loan is subject to the additional conditions precedent that, immediately prior to the making of such Loan: (i) the representations and warranties of Borrowers contained in Article IV of this Agreement, Section 4 of the Security Agreements and Section 5 of the Pledge Agreements shall be true and complete in all material respects on and as of the date of the Loans as if they had been made on such date (except to the extent that such representations and warranties expressly relate to an earlier date or are affected by the consummation of transactions permitted under this Agreement); (ii) the Banks shall have received such additional certificates and documents as they may have reasonably requested prior to such date with respect to the Revolving Loan Documents, the use of the proceeds hereof and compliance with the provisions of the Revolving Loan Documents; (iii) the Banks shall have received reports and certificates which comply with the provisions of Section 6.2; (iv) after giving effect to the Loans to be made on such date, no Event of Default (as specified in Article IX hereof) nor any event which with the giving of notice or expiration of any applicable grace period or both would constitute an Event of Default shall have occurred and be continuing; (v) since the date of this Agreement, there shall have been no material adverse change in the business, assets, operations or condition, financial or otherwise, of Borrowers and the Subsidiaries on a consolidated basis, and (vi) after giving effect to such Loan, the aggregate of the Loan Outstanding shall not exceed the Borrowing Base. Each request for a Loan made by Borrowers hereunder shall constitute a representation and warranty to the Banks that all of the conditions specified in this Section 5.2 have been satisfied in all respects as of the date of each such Loan.

### ARTICLE VI

#### AFFIRMATIVE COVENANTS

Borrowers covenant and agree that, so long as any of the Commitment shall remain available to Borrowers, or any Loans shall remain outstanding and until the principal of and interest on the Notes and all fees and other amounts due hereunder and under the other Revolving Loan Documents shall have been paid in full:

6.1 Financial Statements, etc. Borrowers will furnish or cause to be furnished to each Bank:

(a) within ninety (90) days after the end of each fiscal year of each

Borrower and each Subsidiary (and no later than the date of the filing by Telos Corp. of its Form 10-K for such fiscal year), the balance sheet of Borrowers and the Subsidiaries and the related statements of income, stockholders' equity and cash flow on an unconsolidated, consolidated and consolidating basis as at the end of such year, setting forth in comparative form with respect to such financial statements figures for the previous fiscal year, all in reasonable detail and certified by the Accountants, which opinion shall be unqualified and shall state that such financial statements have been prepared in accordance with GAAP applied on a basis consistent with that of the preceding fiscal year and that the audit by such Accountants in connection with such financial statements has been made in accordance with GAAP;

(b) within forty-five (45) days after the end of each of the first three quarterly accounting periods in each fiscal year of Borrowers (and no later than the date of the filing by Telos Corp. of Form 10-Q for each such quarterly accounting period), Borrowers' and the Subsidiaries' unaudited balance sheet and the related unaudited statements of income, stockholders' equity and cash flow on an unconsolidated, consolidated and consolidating basis for such period and for the period from the beginning of the current fiscal year to the end of such period, including a comparison to budget and a comparison to the figures for the corresponding periods in the immediately preceding fiscal year, all in reasonable detail and certified on behalf of Borrowers by the Executive Officers of Borrowers to be complete and correct in all respects and to have been prepared in accordance with GAAP applied on a basis consistent with that of the preceding fiscal year;

(c) within thirty (30) days after the end of each month, Borrowers' and the Subsidiaries' balance sheets and the related statements of income and cash flow on an unconsolidated, consolidated and consolidating basis for such period and certified on behalf of Borrowers by an Executive Officer to be complete and correct in all respects and to have been prepared in accordance with GAAP applied on a basis consistent with that of the preceding fiscal year;

(d) promptly upon their becoming available, copies of any (i) weekly periodic management reports, budgets, forecasts and other summaries of financial performance of Borrowers and the Subsidiaries as the Agent may reasonably request, and (ii) periodic or special reports filed by a Borrower or a Subsidiary with any federal, state or local governmental agency or authority, and copies of any material notices and other material adverse communications from any federal, state or local governmental agency or authority which specifically relate to a Borrower or a Subsidiary;

(e) promptly upon receipt of any material notice or other communication from any holder of Subordinated Debt, copies of such notice, report or communication;

(f) with reasonable promptness, all management letters from accountants including, without limitation, the annual management letter from the Accountants which shall be provided to each Bank no later than ninety (90) days after the end of each fiscal year;

(g) within forty-five (45) days after the end of each quarter, the contract backlog reports of each Borrower and each Subsidiary;

(h) within two (2) Banking Days of a payment date (whether at maturity or at a date fixed for payment or by declaration, acceleration or otherwise) under the Subordinated Loan Documents confirmation that such payment has been made or if no such payment has been made an explanation of such non-payment; and

(i) within thirty (30) days after the end of each month, a report classifying by age or number of days outstanding all payables of each Borrower and Subsidiary.

6.2 Borrowing Base Reporting. Borrowers will furnish to the Banks the following reports on a bi-weekly basis:

 (a) a report classifying by age or number of days outstanding, by contract, and otherwise categorizing, all Receivables of Borrowers and the Subsidiaries;

(b) a report classifying Inventory of Borrowers and the Subsidiaries by category, location, ownership and value and indicating the amount of inventory under firm orders within one hundred twenty (120) days of the order date;

(c a Borrowing Base Certificate; and

(d) a certificate of a President, Vice President or Chief Financial Officer as to the accuracy of the reports furnished pursuant to subsections (a) through (c) above.

6.3 Legal Existence; Franchises; Compliance with Laws, etc. Each of the Borrowers and the Subsidiaries will (i) maintain its corporate existence and

business; (ii) maintain all properties which are necessary for the conduct of such business, now or hereafter owned, in good repair, working order and condition; (iii) take all actions necessary to maintain and keep in full force and effect its material rights and franchises, including, without limitation, material Franchises and any security clearances which may be required for a Borrower or a Subsidiary to bid for, obtain or perform directly or indirectly, U.S. Government Contracts; and (iv) except as otherwise provided herein, comply in all material respects with all applicable statutes, rules, regulations and orders of, and all applicable restrictions imposed by, all governmental authorities in respect of the conduct of its business and the ownership of its properties. Borrowers will not engage, and will cause the Subsidiaries not to engage, without the express prior written consent of the Banks, in any primary business other than that of providing services with respect to integrating computer systems, computer maintenance, computer software development and related activities.

6.4 Payment of Taxes, etc. Borrowers will pay and discharge, and will cause the Subsidiaries to pay and discharge, promptly as they become due and payable (i) all Taxes imposed upon any of them or their income or upon any of their property or assets, or upon any part thereof, and (ii) all lawful claims of any kind (including claims for labor, materials and supplies) which, if unpaid, might by law become a Lien upon their property; provided, however, that a Borrower or a Subsidiary may defer the payment of any Tax if (a) such Borrower or such Subsidiary, as the case may be, is contesting the imposition or amount of such Tax in good faith and has made adequate reserves on its books therefor, (b) such Borrower or such Subsidiary, as the case may be, is lawfully permitted, in view of such contest, to defer payment and (c) the Agent is satisfied that the deferral of the payment will not impair the Collateral or the Leased Real Property, a Bank's or the Agent's rights to exercise remedies upon an Event of Default or the ability of such Borrower or such Subsidiary to conduct its business. Upon the Agent's request, Borrowers shall provide the Agent with copies of all federal and state corporate income tax filings of Borrowers and the Subsidiaries within two (2) days following the date that such filings were made.

6.5 Payment of Obligations. Borrowers will pay, and will cause the Subsidiaries to pay, promptly when due, or in conformance with customary trade terms, all obligations incident to the conduct of their business other than the Subordinated Debt, which will be paid in accordance with the terms of the Investor Subordination Agreement and this Agreement; provided, however, that a Borrower or a Subsidiary may defer the payment of any obligation if (i) such Borrower or such Subsidiary is contesting the payment of the obligation in good faith and has made adequate reserves on its books therefor, (ii) such Borrower or such Subsidiary is lawfully permitted, in view of such contest, to defer payment and (iii) the Agent is satisfied that the deferral of the payment will not impair the Collateral, the Agent's or a Bank's rights to exercise remedies upon an Event of Default or the ability of such Borrower or such Subsidiary to conduct its business.

6.6 Interest Rate Protection Agreements. Upon request of the Agent, Borrowers shall, within five (5) Banking Days thereof, enter into and cause to remain in effect one or more Interest Rate Protection Agreements satisfying each of the following conditions:

(a) each Interest Rate Protection Agreement shall be in form and substance and in a notional principal amount satisfactory to the Agent;

(b) each Interest Rate Protection Agreement shall terminate not earlier than the Maturity Date and shall provide that Borrowers' payment amount, if any, be calculated on the basis of a fixed rate of interest on the notional principal amount thereof; and

(c) the counterpart to each Interest Rate Protection Agreement shall be a bank or other financial institution satisfactory to the Agent.

6.7 Further Assurances. From time to time hereafter, Borrowers will execute and deliver, or will cause to be executed and delivered, at Borrowers' sole expense, such additional instruments, certificates or documents, and will take all such related actions as any Bank or the Agent may reasonably request for the purposes of implementing or effectuating the provisions of the Revolving Loan Documents.

6.8 Compliance Certificates. Within forty-five (45) days after the end of each calendar quarter, Borrowers will deliver to the Agent a certificate, signed by an Executive Officer substantially in the form attached hereto as Exhibit E (a "Compliance Certificate"), stating that a review of the activities of Borrowers and the Subsidiaries during such reporting period has been made under their supervision and that no Event of Default or condition or event which, with notice or lapse of time, or both, would constitute an Event of Default has occurred, or, if such has occurred, specifying the nature and status thereof.

6.9 Notice of Defaults, Disputes and Other Matters. Borrowers will execute and deliver to the Agent a certificate signed by an Executive Officer, specifying the nature and status (including a discussion of what action a Borrower has taken, is taking or proposes to take with respect thereto) not later than fifteen (15) days (or such other time period as is specified in any of the following clauses) after any officer of a Borrower becomes aware, or should have become aware, of the occurrence of the following events (except that in the case of any event described in clauses (j), (k) and (l) of this Section 6.9, such certificate shall be so executed and delivered promptly upon any such officer so becoming aware of such event):

(a) a Borrower or any of the Subsidiaries shall have changed its respective corporate name or shall do business under any name other than its respective name as of the date hereof; or

(b) a Borrower or any of the Subsidiaries shall have changed its principal place of business or its chief executive office or shall have opened any new principal place of business; or

(c) a Borrower or any of the Subsidiaries shall have changed its fiscal year; or

(d) a Borrower or any of the Subsidiaries shall have opened or closed any material place of business; or

(e) any strike or walkout shall have occurred relating to any office or facility owned or leased by a Borrower or any of the Subsidiaries or any labor contract to which a Borrower or any of the Subsidiaries is a party or by which it is bound expires; or

(f) any citations, order to show cause, or other legal process or order, or protest, or reconsideration affecting a Franchise or Consent held by a Borrower or any of the Subsidiaries or directing a Borrower or any of the Subsidiaries to become a party to or to appear at any proceeding or hearing by or before any governmental instrumentality (including, without limitation, any federal, state or local municipality or other instrumentality which shall have granted to a Borrower or any of the Subsidiaries a Consent or Franchise) which might result in any of the foregoing, either individually or in the aggregate, being materially adverse to a Borrower or any of the Subsidiaries, including with such notice a copy of any such citation, order to show cause or other legal process or order; or

(g) any (i) refusal or failure by any governmental agency or other instrumentality to renew or extend any material Consent or Franchise with respect to the business of a Borrower or any of the Subsidiaries, (ii) proposed denial, threatened denial, abandonment or proposed or actual revocation, termination or materially adverse modification of any Consent or Franchise by any Person, or (iii) threats, notices, or requests by any Person with respect to any of the foregoing or with respect to any proceeding or hearing relating to the foregoing, which might result in any of the foregoing, either individually or in the aggregate, being materially adverse to a Borrower or any of the Subsidiaries; or

(h) any dispute concerning, or any threatened non-renewal or modification of, any lease for real or personal property to which a Borrower or any of the Subsidiaries is a party, if such dispute, threatened nonrenewal or modification, either individually or in the aggregate, could have a material adverse effect upon a Borrower or any of the Subsidiaries; or

(i) a Borrower or any of the Subsidiaries shall become a party to one or more suits, actions or proceedings which could have a material adverse effect on the business, assets, operations or condition, financial or otherwise, of such Borrower or such Subsidiary; or

(j) any condition or event which constitutes or which, with notice or lapse of time, or both, would constitute an Event of Default shall have occurred; or

(k) the occurrence of any event which constitutes, or with notice or lapse of time or both would constitute, a default or an event of default under any contractual obligation of a Borrower or any of the Subsidiaries, which, if adversely determined, could, either individually or in the aggregate, have a material adverse effect on the business, assets, operations or condition, financial or otherwise, of a Borrower or any of the Subsidiaries; or

(1) an ERISA Termination Event or a material "prohibited transaction" (as such term is defined in Section 4975 of the Code) with respect to any Plan has occurred, or any officer of a Borrower or any of the Subsidiaries has knowledge of the PBGC's intention to terminate or have a trustee appointed to administer any Plan; or

(m) as soon as possible and in any event (A) within thirty (30) days after a Borrower or any of the Subsidiaries or any ERISA Affiliate thereof knows that any ERISA Termination Event described in clause (a) of the definition of ERISA Termination Event with respect to any Plan of a Borrower or any of the Subsidiaries has occurred and (B) within ten (10) days after a Borrower or any of the Subsidiaries or any ERISA Affiliate thereof knows that any other ERISA Termination Event with respect to any Plan of a Borrower or any of the Subsidiaries or any ERISA Affiliate thereof has occurred, a statement of the Chief Financial Officers of Borrowers describing such ERISA Termination Event and the action, if any, which a Borrower or any of the Subsidiaries or such ERISA Affiliate proposes to take with respect thereto; or

(n) promptly and in any event within ten (10) days after receipt thereof by a Borrower or any of the Subsidiaries or any ERISA Affiliate from the PBGC, copies of each notice received by a Borrower or any of the Subsidiaries or such ERISA Affiliate of the PBGC's intention to terminate any Plan of such Borrower or any of the Subsidiaries or such ERISA Affiliate or to have a trustee appointed to administer any Plan of such Borrower or any of the Subsidiaries or such ERISA Affiliate; or

 (o) promptly and in any event within thirty (30) days after the reasonable request of the Agent, copies of each Schedule B (Actuarial Information) to the annual report (Form 5500 Series) with respect to each Plan; or

(p) promptly and in any event within ten (10) days after receipt thereof by a Borrower or any of the Subsidiaries or any ERISA Affiliate thereof from the sponsor of a Plan which is a multiemployer plan, a copy of each notice received by a Borrower or any of the Subsidiaries or such ERISA Affiliate concerning the imposition or amount of withdrawal liability pursuant to Section 4202 of ERISA; or

(q) promptly after a Borrower or any of the Subsidiaries obtains knowledge of the commencement thereof, notice of all actions, suits and proceedings before any court or governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, affecting a Borrower or any of the Subsidiaries of the type described in Section 4.16; or

(r) any condition shall exist which has resulted in or which is likely, in the reasonable judgment of Borrowers, to result in a material adverse change in the business, assets, operations or condition, financial or otherwise, of a Borrower or any of the Subsidiaries; or

(s) there shall have occurred an event of default under any of the Subordinated Loan Documents.

6.10 Inspection and Other Information. Each Bank, and such persons as such Bank may designate, may visit and inspect any of the properties of Borrowers and the Subsidiaries, examine their books of account, take extracts therefrom and discuss the affairs, finances and accounts of Borrowers and the Subsidiaries with their officers, employees and public accountants (and by this provision Borrowers authorize the Accountants to discuss with any such Bank and such persons their finances and accounts), at such reasonable times during business hours and as often as such Bank may reasonably desire.

6.11 Compliance with Environmental, OSHA and ERISA Laws. Borrowers will comply, and will cause the Subsidiaries to comply, in all material respects with all applicable statutes, rules, regulations and orders of all governmental authorities with respect to the conduct of their businesses and the ownership of their properties, including, without limitation, (i) all applicable statutes, rules, regulations and orders relating to environmental protection and pollution control, (ii) the Occupational Safety and Health Act of 1970, as amended, and (iii) ERISA.

6.12 Notices of Events Involving Securities. Borrowers will give notice to each Bank with a copy to the Agent within two (2) days after Borrowers shall have filed with the Commission or with any national securities exchange (as defined in the Exchange Act) either an application to register any securities of a Borrower or a Subsidiary pursuant to Section 12 of the Exchange Act or a registration statement under Section 5 of the Securities Act relating to any securities of a Borrower or a Subsidiary.

6.13 Protection of the Collateral. Borrowers shall, at their sole expense, defend the Collateral and the Leased Real Property against all claims and demands of all other Persons at any time claiming any Lien on or interest therein. Other than sales by Borrowers of Inventory in the ordinary course of business and in compliance with the Security Documents, Borrowers shall not and shall cause each Subsidiary not to transfer, sell or assign any interest in the Collateral or the Leased Real Property or permit any Lien to be created or remain thereon, other than the Liens expressly created by this Agreement and the Security Documents.

6.14 Press Releases; Notice of Press Releases. Borrowers shall provide the Agent, within one (1) day upon their becoming available, all press releases and other public announcements of Borrowers and the Subsidiaries. Without limiting the generality of the foregoing, Borrowers shall consult with the Agent prior to the release of any press release or other public announcement of Borrowers and the Subsidiaries that mentions the Agent or any Affiliate, or relates to any transaction involving the Agent or any Affiliate or any role or participation by the Agent or any of its Affiliates (whether or not named) in any transaction, without, in each case, the Agent's express prior written consent, such consent not to be unreasonably withheld.

6.15 Bank Accounts. Borrowers shall maintain, and shall cause the Subsidiaries to maintain, all bank accounts (or other similar accounts utilized for the purpose of investing short-term cash holdings) having balances in excess of \$25,000 under the names of Borrowers only and only with NationsBank (the "Deposit Accounts"). Borrowers shall not and shall cause each Subsidiary not to maintain aggregate balances exceeding \$50,000 in bank accounts other than the Deposit Accounts. From and after the date hereof, Borrowers shall notify the Agent of the opening by either Borrower or any Subsidiary of any bank accounts.

6.16 Management of Borrowers. Borrowers shall provide the Agent with detailed information regarding the management structure of Borrowers and the Subsidiaries, and shall advise the Agent promptly of any future changes to be made.

6.17 Landlords'/Mortgagees' Waivers. Within thirty (30) days after the Closing Date, Borrowers shall provide to the Agent a waiver, in form and substance acceptable to the Agent, executed by the landlord of Telos Corp.'s principal offices located at 19886 Ashburn Road, Ashburn, Virginia 20147. Within thirty (30) days after either Borrower or any Subsidiary enters into any lease for Leased Real Property, Borrowers shall provide to the Agent waivers, in form and substance acceptable to the Agent, executed by the landlords of such property. Such waivers shall include a consent to each Assignment of the Tenant's Interest Under Lease and shall assure the Agent of its ability to gain access to such premises and remove the Collateral without interference by the landlord, even if Borrowers are in default under any leases or mortgages (or deeds of trust) affecting such premises.

6.18 enterWorks Subordination Agreement. Within thirty (30) days after the Closing Date, Borrowers shall deliver to the Agent the enterWorks Subordination Agreement which is duly executed and delivered by each holder of the enterWorks Notes.

6.19 Bringdown of Schedules. Within thirty (30) days after the Closing Date, Borrowers shall deliver to the Agent updated Schedules 4.12 and 4.28, which shall set forth true and correct information as of the date of the delivery of such Schedules.

#### ARTICLE VII

## NEGATIVE COVENANTS

Borrowers covenant and agree that, so long as the Commitment shall remain available to Borrowers or any Loans shall remain outstanding, and until the principal of and interest on the Notes and all fees and other amounts due hereunder and under the other Revolving Loan Documents shall have been paid in full to the satisfaction of the Agent:

7.1 Indebtedness. Borrowers will not, and will cause the Subsidiaries not to, create, incur, assume or become or remain liable in respect of any Indebtedness, except:

- (a) the Notes;
- (b) the Subordinated Debt;

(c) Indebtedness pertaining to Capital Expenditures in an aggregate amount notexceeding \$250,000 during any fiscal year;

(d) Indebtedness incurred to repay in full or to refinance all amounts due the Banks under the Revolving Loan Documents; and

(e) Guaranties of Subsidiaries.

7.2 Liens. Borrowers will not, and will cause the Subsidiaries not to, directly or indirectly, create, incur, assume or suffer to exist, any Lien with respect to any property or asset now owned or hereafter acquired by Borrowers or the Subsidiaries including, without limitation, the Collateral and the Leased Real Property, except:

(a) Liens on assets or property of Borrowers or the Subsidiaries existing on the date hereof and described in Schedule 7.2(a) which are to remain outstanding, but no extension or renewal thereof;

(b) Liens securing the payment of taxes, assessments and governmental charges or levies, provided that (i) a Borrower or a Subsidiary is contesting the imposition or amount of such tax in good faith and has made adequate reserves on its books therefor, (ii)such Borrower or such Subsidiary is lawfully permitted, in view of such contest, to defer payment and (iii) the Agent is satisfied that the deferral of the payment will not impair the Collateral or the Leased Real Property, any Bank's or the Agent's right to exercise remedies upon an Event ofDefault or the ability of such Borrower or such Subsidiary to conduct its business; (c) Liens incurred in connection with the security interest in the Collateral created hereby and in certain Collateral and certain other assets of Telos Corp. under the Security Documents;

(d) Liens incurred in connection and simultaneously with the acquisition of capital assets as provided in Section 7.1(c) hereof not to exceed \$250,000 at any one time;

(e) Liens on securities pledged as collateral to secure performance bonds or bid bonds obtained in the ordinary course of business, provided that the fair market value of securities so pledged shall not exceed at any time \$250,000 in the aggregate;

(f) Permitted Liens; and

(g) the PP&E Lien, as defined in the Investor Subordination Agreement.

7.3 Corporate Restrictions. Borrowers will not, and will cause the Subsidiaries not to, directly or indirectly:

(a) amend or otherwise change their Articles of Incorporation or By-Laws or the terms of any preferred stock of a Borrower, including the Telos Corp. Preferred Stock, in a manner that would reasonably be likely to have a material adverse effect on the business, assets, projects, operations or conditions, financial or otherwise, of a Borrower or a Subsidiary; or

(b) issue, sell, pledge, or authorize the issuance of any preferred stock (other than the issuance by Telos Corp. of preferred stock having terms substantially equal to that of the Telos Corp. Preferred Stock), any redeemable or exchangeable security or any security convertible into any of the foregoing or any right to acquire any of the foregoing; or

(c) other than with respect to the Public Preferred Stock, reclassify, combine, split or subdivide any of its capital stock; or

(d) other than dividends payable by Telos Sub or a Subsidiary to its corporate parent, make any payment of cash dividends, or any other cash distribution, whether direct or indirect, on or on account of any class of stock of Borrowers or any Subsidiary now or hereafter outstanding; provided, however, that Borrowers may make payments of dividends on the Telos Corp. Preferred Stock if, after giving effect to such payments, Borrowers and the Subsidiaries are in full compliance with the financial covenants specified in Article VIII and there exists no Event of Default; provided, however, that Borrowers hereby agree not to seek and the Banks shall not grant any waiver to permit the payment of cash dividends on Telos Corp. common stock unless Borrowers simultaneously seek and the Banks also agree to simultaneously grant a waiver to permit the cash payment under the Warrant Agreement required as a result of the payment of such dividend on the Telos Corp. common stock as described in the Warrant Agreement.

(e) other than (i) any repurchase of the Public Preferred Stock, (ii) any redemption in accordance with the provisions of Telos Corp. Preferred Stock and (iii) any redemption or purchase of enterWorks Stock or Telos Corp. common stock from departing employees in a total amount not to exceed \$250,000 in any fiscal year, redeem, purchase or otherwise acquire, directly or indirectly, any shares of any class of stock of Telos Corp. now or hereafter outstanding or of any warrants or rights to purchase any such stock (including, without limitation, the repurchase of any such stock or warrant or any refund of the purchase price thereof in connection with the exercise by the holder thereof of any right of recision or similar remedies with respect thereto); provided, however, that any repurchase or redemption permitted by this clause (e) shall be made only if, after giving effect to such repurchase or redemption, Borrowers and the Subsidiaries are in full compliance with the financial covenants specified in Article VIII and there exists no Event of Default; or

(f) make any changes other than in the ordinary course of business and in a manner consistent with past practice with respect to accounting policies or procedures, other than those required by GAAP; or

(g) change its fiscal year; or

(h) agree or commit to do any of things listed in Sections 7.3(a) through (g) above.

7.4 Loans and Investments. Borrowers will not make or permit to remain outstanding any loan or advance to, or guarantee or endorse (except as a result of endorsing negotiable instruments for deposit or collection in the ordinary course of business) or otherwise assume or remain liable with respect to any obligation of, or make or own any investment in, or acquire (except in the ordinary course of business) the properties or assets of, any Person, and will cause the Subsidiaries not to engage in any of the foregoing activities except:

(a) the presently outstanding investments and loans referred to in Schedule 7.4 attached hereto;

## (b) transactions permitted by Section 7.1(c); and

(c) loans to employees of Borrowers or Subsidiaries in an aggregate outstanding amount not to exceed \$150,000 at any time; and

(d) the Permitted Investments.

7.5 Leases. Borrowers will not, and will cause the Subsidiaries not to, enter into any Capitalized Lease, except as otherwise permitted under Section 7.1(c).

7.6 Consolidation; Sale of Assets. Borrowers will not make any substantial change in the nature of their businesses, and will not sell, lease or otherwise dispose of any part of their properties or assets (other than in the ordinary course of business) or merge or consolidate with any other Person and will cause the Subsidiaries not to do any of the foregoing.

7.7 Acquisition of Stock and Other Interests; Creation of Subsidiaries. Except as otherwise permitted by Section 7.3(e), Borrowers will not, and will cause the Subsidiaries not to, acquire or hold or make any offer to acquire, in any manner, any capital stock of a corporation or any interest whatsoever in any partnership, joint venture, or other entity. Borrowers will not, and will cause the Subsidiaries not to, create any corporate subsidiaries, nor will they participate as a partner or joint venturer in any partnership or joint venture to the extent such partnership or joint venture will obligate Borrowers or the Subsidiaries to contribute funds if an Event of Default would result hereunder; provided, however, that this provision shall not restrict the ability of Borrowers to bid contracts jointly with other government contractors.

7.8 Amendment of Other Documents. Borrowers will not, without the express prior written consent of the Agent, consent or agree to any amendment of, modification of or waiver with respect to the Subordinated Loan Documents, or any other agreements relating to the foregoing. Borrowers will not enter into any agreement or instrument relating to any Indebtedness which in any way limits or restricts a Borrower's ability to perform its obligations under the Revolving Loan Documents.

7.9 Sale and Leasebacks. Borrowers will not, and will cause the Subsidiaries not to, enter into any arrangement, directly or indirectly, with any Person whereby they shall sell or transfer any property, real or personal, used or useful in their business, whether now owned or hereafter acquired, and thereafter rent or lease such property or other property which they intend to use for substantially the same purpose or purposes.

7.10 Transactions with Affiliates. Borrowers will not, and will cause the Subsidiaries not to, directly or indirectly, enter into any lease or other transaction with any Affiliate of a Borrower or Subsidiary or any shareholders of a Borrower, Subsidiary or any Affiliate of any shareholder of a Borrower or a Subsidiary unless such Borrower shall have reasonably determined such transaction is on terms that are no less favorable to such Borrower or Subsidiary than those which might be obtained at the time from Persons who are not Affiliates.

7.11 ERISA. Borrowers will not, and shall cause the Subsidiaries not to, permit any employee pension benefit plan, as that term is defined in ERISA, maintained by either of them to (a) engage in any "prohibited transaction" as that term is defined in Section 4975 of the Code, (b) incur any "accumulated funding deficiency" as that term is defined in ERISA, whether or not waived, or (c) terminate in any manner which could result in the imposition of a lien or encumbrance on the assets of a Borrower or any owned Subsidiary pursuant to Section 4068 of ERISA.

7.12 Observance of Subordinated Loan Documents. Borrowers will not make, or cause or permit to be made: (i) any payments in respect of Subordinated Debt in contravention of the Investor Subordination Agreement, the enterWorks Subordination Agreement or other agreement evidencing such Subordinated Debt; (ii) any prepayment in respect of principal of, or premium, if any, in respect of, Subordinated Debt; (iii) any payment of cash interest in respect of the Subordinated Debt; provided, however, that, except as otherwise permitted in the Investor Subordination Agreement or the enterWorks Subordination Agreement, Borrowers may make payment of accrued but unpaid interest in cash described in clause (iii) above (including interest accrued on due but unpaid interest) on regularly scheduled interest payment dates; or (iv) any payment which would result in the defeasance of any instrument under which the Subordinated Debt is outstanding; nor will Borrowers amend, modify or change in any manner any of such subordination provisions without the prior written consent of the Bank.

7.13 Government Contracts. Borrowers will not, and shall cause the Subsidiaries not to, take any action that would result in: the suspension or debarment from U.S. Government contracting; default or termination of any current U.S. Government Contract; violation of any of the statutes referred to in Section 4.27 or Section 4.34; suspension or extinguishment of the right of the Agent to receive payments pursuant to any assignment of any U.S. Government Contracts made pursuant to the Revolving Loan Documents; any assignment to any party other than the Agent of the right to payment under any contract with respect to which Receivables have been or are to be assigned to the Agent hereunder; or termination or suspension of any facility security clearance or personnel security clearance held by a Borrower or a Subsidiary.

7.14 Use of Proceeds. Borrowers will not use the proceeds of the Loans for any purposes other than those set forth in, and in accordance with the provisions of, Section 4.24.

7.15 Capital Expenditures. Borrowers will not, directly or indirectly, make Capital Expenditures in any fiscal year in excess of \$2,700,000, or such other amount approved by the Bank in advance upon submission by Borrowers of the annual budget for capital expenditures which submission shall be in writing no later than February 15 of each fiscal year, which approval shall not be unreasonably withheld.

7.16 Negative Pledge. Borrowers will not, and will not permit any Subsidiary to, enter into any security agreement, pledge agreement, collateral assignment, mortgage or other agreement, instrument or document (including, without limitation, financing statements) which gives rise to, creates or perfects a Lien against the Collateral or the Pledged Stock (other than the Security Documents) or against the enterWorks Stock. Borrowers will not, and will not permit any Subsidiary to, create, incur, assume or permit to exist any Lien, or any other encumbrance against, on or with respect to the enterWorks Stock.

7.17 Restrictions on Intercompany Transactions. Borrowers will not, and will not permit any Subsidiary to, declare, order, pay, make, advance or set apart any sum or property to any Subsidiary or joint venture to which a Borrower or a Subsidiary is a party without the prior written consent of the Agent; provided, however, that Borrowers may make advances (a) to Telos International Corp. in an amount not to exceed \$500,000 in any fiscal year and (b) to Telos Field Engineering for Capital Expenditures.

### ARTICLE VIII

#### FINANCIAL COVENANTS

Borrowers covenant and agree that for so long as the Commitment shall remain available to Borrowers, or any Loans shall remain outstanding, and until the principal of and interest on the Notes and all fees and other amounts due hereunder and under the other Revolving Loan Documents shall have been paid in full to the satisfaction of the Agent:

#### 8.1 Fixed Charge Coverage.

On a consolidated basis, the ratio of the sum of pretax income, depreciation, interest expense and amortization minus cash taxes paid and unfinanced capital expenditures (including capitalized software development costs) (the "Numerator"), divided by the sum of amounts paid or payable for debt principal repayment, interest expense after giving effect to interest payments, if any, received under any Interest Rate Protection Agreement, and capitalized lease payments (the "Denominator") shall not be less than 1.25:1 for any calendar quarter. This covenant shall be tested on the last day of each calendar quarter.

## 8.2 Minimum Tangible Capital Funds.

On a consolidated basis, Tangible Capital Funds shall not be less than the following during any of the periods described below:

Period	Required Amount
Closing Date through September 29, 1997	\$6,500,000
September 30, 1997 through December 30, 1997	\$7,250,000
December 31, 1997 through December 30, 1998	\$9,500,000
December 31, 1998 through the Maturity Date	\$14,000,000

On a consolidated basis, Net Worth shall not be less than the following during any of the periods described below:

Period	Required Amount
Closing Date through September 29, 1997	\$6,500,000
September 30, 1997 through December 30, 1997	\$7,250,000
December 31, 1997 through December 30, 1998	\$9,500,000
December 31, 1998 through the Maturity Date	\$12,500,000

8.4 Leverage Ratio.

On a consolidated basis, the ratio of Total Liabilities minus Subordinated Debt to Net Worth plus Subordinated Debt will not at any time exceed the following during any of the periods described below:

Period	Required Ratio
Closing Date through December 30, 1997	4.0:1
December 31, 1997 through December 30, 1998	3.2:1
December 31, 1998 through the Maturity Date	3.0:1

8.5 Borrower's Operating Income.

On a consolidated basis, the sum of Borrowers' net income, depreciation and amortization (excluding amounts reported as capitalized lease amortization but which are actually cash rental expenses paid) shall not result in a net loss for any calendar quarter.

8.6 enterWorks Financial Performance.

(a) At any time during fiscal year 1997, enterWorks shall not incur a cumulative operating loss greater than \$4,000,000; provided, however, that enterWorks may incur a cumulative operating loss in excess of \$4,000,000 in the event, but only to the extent, enterWorks obtains additional equity or financing, which amount shall be determined by the Agent in its sole and absolute discretion.

(b) At any time after fiscal year 1997, enterWorks shall not incur a cumulative operating loss.

(c) At any time during fiscal year 1997, enterWorks shall not incur capitalized product and software development costs not otherwise included in the calculation of operating loss for purposes of Section 8.6(a) in an aggregate amount in excess of \$2,000,000; provided, however, that enterWorks may incur such costs in excess of \$2,000,000 in the event, but only to the extent, enterWorks obtains additional equity or financing, which amount shall be determined by the Agent in its sole and absolute discretion.

#### ARTICLE IX

#### DEFAULTS; REMEDIES

9.1 Events of Default; Acceleration. Each of the following shall constitute an "Event of Default" under this Agreement:

(a) if Borrowers fail to pay (x) the principal of or premium, if any, on any Note, when the same becomes due and payable, whether at the maturity thereof, on a date fixed for payment or for a prepayment, or otherwise, (y) the interest on any Note or (z) a fee or other amount payable hereunder, when and as the same becomes due and payable, and such failure shall have continued for five (5) days; or

(b) if Borrowers shall default in the performance or compliance with any term contained in Sections 6.3, 6.4, 6.5, 6.6 or 6.11 hereof, and such

default shall not have been remedied within ten (10) days of the occurrence of such default; or

(c) if Borrowers default in the performance of or compliance with any term contained in Articles VI (other than those listed in clauses (a) or(b) hereof), VII or VIII; or

(d) if Borrowers shall default in the performance of or compliance with any agreement contained herein other than those referred to above in this Article IX and such default shall not have been remedied within thirty (30) days after written notice thereof shall have been given to Borrowers by the Agent; or

(e) if a Borrower or a Subsidiary shall default in the performance of or compliance with any agreement contained in the other Revolving Loan Documents or in any of the Subordinated Loan Documents, or in the performance of or compliance with any material term contained in any other written agreement in respect of Indebtedness between a Borrower or a Subsidiary or the Banks (or the Agent on their behalf), and such default shall continue for more than the period of grace, if any, specified therein and shall not have been waived pursuant thereto; or

(f) if any representation or warranty made by Borrowers herein, in the other Revolving Loan Documents or pursuant hereto or thereto shall prove to have been false or incorrect in any material respect on the date as of which made; or

(g) if a Borrower or a Subsidiary shall default in any payment due on any Indebtedness in respect of borrowed money with an unpaid principal amount in excess of \$250,000, any Interest Rate Protection Agreement, any Capitalized Lease or the deferred purchase price of property, and any such default shall continue for more than the period of grace, if any, specified therein and shall not have been waived pursuant thereto; or

(h) if a Borrower or a Subsidiary shall default in payment or performance of any material obligation (except Indebtedness, which is covered in Section 9.1(g) above), whether now or hereafter incurred, and such default shall continue for more than the period of grace, if any, specified in the agreement or other documents setting forth the terms of such obligation, or shall not have been waived pursuant thereto, and in the reasonable judgment of the Agent, either individually or together with any other defaults hereunder, jeopardizes or could reasonably be expected to jeopardize repayment of any Note, or consummation of the transactions contemplated in the Revolving Loan Documents or the Subordinated Loan Documents; or

(i) if any action, proceeding, regulation or legislation shall have been instituted, threatened or proposed before any court, governmental agency or legislative body to enjoin, restrain, or prohibit, or to obtain damages in respect of, or which is related to or arises out of the Revolving Loan Documents, the Subordinated Loan Documents, or the transactions contemplated hereby or thereby; provided, however, that the institution or threat of such an action or proceeding shall not constitute an Event of Default hereunder if (i) independent counsel retained by Borrowers shall have delivered to the Banks an opinion stating that there is no material likelihood that such action or proceeding will be adversely determined against a Borrower or a Subsidiary or (ii) the outcome of the proceeding, if adversely determined, will not have a material and adverse effect on the business, condition, financial or otherwise, or operations of a Borrower or a Subsidiary or the transactions contemplated by the Revolving Loan Documents or the Subordinated Loan Documents; or

(j) if a Borrower or a Subsidiary shall discontinue its primary business, or shall make a general assignment for the benefit of creditors, or shall fail generally to pay its debts as such debts become due, or shall apply for or consent to the appointment of or taking possession by a trustee, receiver or liquidator (or other similar official) of a Borrower or a Subsidiary or any substantial part of the property of a Borrower or a Subsidiary, or if there shall occur any commencement by any Borrower or Subsidiary of a voluntary case or proceeding under any applicable federal or state bankruptcy, insolvency, reorganization or other similar law or of any other case or proceeding to be adjudicated a bankrupt or insolvent, or the consent by any

Borrower or Subsidiary to the entry of a decree or order for relief in respect of such Borrower or Subsidiary in an involuntary case or proceeding under any applicable federal or state bankruptcy, insolvency reorganization or other similar law or the commencement of any bankruptcy or insolvency case or proceeding against any Borrower or Subsidiary, or the filing by any Borrower or Subsidiary of a petition or answer or consent seeking reorganization or relief under any applicable federal or state law, or the consent by any Borrower or Subsidiary to the filing of such petition or to the appointment of or taking possession by a custodian receiver, liquidator, assignee, trustee, sequestrator or similar official of such Borrower or Subsidiary or of any substantial part of such Borrower or Subsidiary, or the making of an assignment for the benefit of creditors, or the admission by any Borrower or Subsidiary in writing of its inability to pay its debts generally as they become due, or the taking of corporate action by any Borrower or Subsidiary in furtherance of any such action; or (k) if there shall occur any refusal or failure by the U.S. Government, any state, or any agency or instrumentality thereof to renew or extend any material Franchise or Consent held by a Borrower or a Subsidiary (provided, however, that any refusal or failure to renew or extend a Consent or Franchise required for performance by a Borrower or a Subsidiary of U.S. Government Contracts shall be considered material) or any denial, forfeiture or revocations by any governmental authority or any authorization required by law or the expiration without renewal of any such authorization, and such events, either individually or in the aggregate, jeopardize, or could reasonably be expected to jeopardize, repayment of the Notes or the continuation of the business of a Borrower or a Subsidiary as it has been conducted in the past; or

(1) (i) if there shall be commenced against a Borrower or a Subsidiary a case under the federal bankruptcy laws, as now or hereafter constituted, or any other applicable federal or state bankruptcy, insolvency or other similar law, and within sixty (60) days of such commencement such case shall not have been dismissed or all orders or proceedings thereunder affecting the operations or the business of a Borrower or a Subsidiary stayed, or if the stay of any such order or proceeding shall thereafter be set aside, or (ii) if, within sixty (60) days after the entry of a decree appointing a trustee, receiver or liquidator (or other similar official) of a Borrower or any substantial part of the property of a Borrower or a Subsidiary such appointment shall not have been vacated; or

(m) a final judgment which, with other outstanding final judgments against a Borrower or a Subsidiary, exceeds an aggregate of \$200,000 shall be rendered against a Borrower or a Subsidiary and if, within sixty (60) days after entry thereof, such judgment shall not have been discharged or execution thereof stayed pending appeal, or if, within sixty (60) days after the expiration of any such stay, such judgment shall not have been discharged or a warrant of attachment or execution or similar process shall be issued or levied against any property of a Borrower or a Subsidiary and if, within three (3) Banking Days after the issue or levy thereof, such warrant or process shall not have been discharged or stayed pending appeal (whether by action of a court, by agreement or otherwise), or if, within three (3) days after the expiration of any such stay, such warrant or process shall not have been discharged; or

(n) if a Borrower or a Subsidiary loses, fails to keep in force, suffers the termination or revocation of or terminates, forfeits or suffers an amendment to any Franchise or Consent at any time held by it which would have a material adverse effect on its financial condition; or

(o) if a Change of Control shall occur; or

(p) if there shall occur (i) any suspension, termination (other than a termination for convenience of the U.S. Government) or revocation of any kind (exclusive of expiration and non-renewal) of any Government Contract or any commercial contract of a Borrower or any Subsidiary with a remaining contract value of \$10,000,000 or more; (ii) any investigation of, or sanctions (including, without limitation, fines, penalties or forfeitures) against, a Borrower or any Subsidiary which has, or is likely to have, a materially adverse effect on the business or operations of Borrowers and the Subsidiaries taken as a whole, including, without limitation, any final decision subject to any applicable disputes clause or unilateral Government Contract modification assessing a material penalty or material damages; any assertion of a material claim based on asserted violations of Cost Accounting Standards ("CAS") or defective pricing; any notice of a proposed material disallowance of indirect cost claims in excess of reserves therefor; any subpoena or notice signifying the initiation of an investigation by Government investigative or enforcement agency other than normal audits in the ordinary course of business; any notice from a Government that any material cost or claim contained in any invoice or request for progress payment has been disallowed; any other notice from the cognizant Government contracting officer alleging the failure by a Borrower or any Subsidiary to provide goods or services fully in conformity to, or otherwise to comply in any material way with, any applicable Government Contract provision, specification or regulation; or (iii) the initiation of a debarment or suspension proceeding by the Government against a Borrower or any Subsidiary; or

(q) if at any time any of the following individuals, or substitutions for such individuals satisfactory to the Agent in its sole and absolute discretion, arenot in the respective positions listed below:

John B. Wood

President and Chief Executive Officer

Lorenzo Tellez

Vice President, Chief Financial Officer and Treasurer; or

(r) If the Subordinated Debt or preferred stock of any Borrower or any Subsidiary is the subject of any restructuring, refinancing, modification or substitution without the prior written consentof the Agent; or

(s) If any action, suit, litigation or other proceeding at law or in

equity is commenced or threatened (i) seeking to enjoin or to unwind the transactions contemplated by that certain Settlement Agreement by and among, inter alia, John R.C. Porter and C3 Investors, L.P. dated as of September 27, 1993 or any document or agreement referenced therein, (ii) seeking to dispute the legality, permissibility oradvisability of the settlement of the cases styled Fred Knoll, et al. v. C3, Inc., et al., Case No. 931119026, CE163830 in the Circuit Court for Baltimore City and Knoll v. Porter, Civil Case No. 93125044 in the CircuitCourt for Baltimore City (the "Litigation"), or (iii) involving any of the issues which were raised by or in connection with the Litigation; or

(t) If any of the provisions of the Standstill Agreement among John R.C. Porter, Fred Knoll and Alfredo Frohlich, dated as of September 27, 1993, are breached or such agreement is no longer in full force and effect; or

(u) If all or any part of the Litigation is reinstituted by any party with respect to any matter arising out of the Litigation;

then, and at any time thereafter, if any such Event of Default shall then be continuing, either or both of the following actions may be taken. The Agent shall (i) declare the principal of and accrued interest in respect of Notes to be forthwith due and payable, whereupon the principal of and accrued interest in respect of the Notes shall become forthwith due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by Borrowers, and/or (ii) declare the Commitment terminated, whereupon the Commitment of the Banks to make Loans hereunder shall forthwith terminate without any other notice of any kind; provided, however, that, notwithstanding the above, if there shall occur an Event of Default under clauses (j), (l), (o), (s), (t), or (u) above, then any and all of the Obligations shall be immediately due and payable without presentment, demand, protest or other notice of any kind by the Agent, all of which are hereby expressly waived by Borrowers.

9.2 Remedies on Default, etc. In case any one or more  $\ensuremath{\mathsf{Events}}$  of Default shall occur and are continuing, the Agent shall be entitled to proceed to protect and enforce the rights of the Banks by an action at law, suit in equity or other appropriate proceeding, whether for the specific performance of any agreement contained herein or in any Revolving Loan Documents, or for an injunction against a violation of any of the terms hereof or thereof, or in aid of the exercise of any power granted hereby or thereby or by law. Without limiting the generality of the foregoing, the Agent shall be entitled (i) to exercise all of the rights and remedies of a secured party under the U.C.C. with respect to the Collateral including, without limitation, those described in the Security Documents and (ii) to exercise all rights and remedies of a creditor available at law and equity with respect to the Leased Real Property. In case of a default in the payment of any principal of or interest on the Notes, or in the payment of any fee or other amount due hereunder, Borrowers will pay to the Banks such further amount as shall be sufficient to cover the cost and expense of collection, including, without limitation, reasonable attorneys' fees, including allocated cost of in-house counsel expenses and disbursements. No course of dealing and no delay on the part of the Banks or the Agent in exercising any right shall operate as a waiver thereof or otherwise prejudice the Agent or any Bank's rights. No right conferred by any Revolving Loan Document upon the Agent or any of the Banks shall be exclusive of any other right referred to therein or now or hereafter available at law, in equity, by statute or otherwise.

9.3 Waiver of Stay or Extension Laws. Each Borrower covenants (to the extent that it may lawfully do so) that it will not at any time insist upon, or plead, or in any manner whatsoever claim or take the benefit or advantage of, any stay or extension law wherever enacted, now or at any time hereafter in force, which may affect the covenants or the performance of any of the Revolving Loan Documents; and each Borrower (to the extent that it may lawfully do so) hereby expressly waives all benefit or advantage of any such law and covenants that it will not hinder, delay or impede the execution of any power herein granted to the Banks or the Agent, but will suffer and permit the execution of every such power as though no such law had been enacted.

9.4 Banks May File Proofs of Claim. In case of the pendency of any receivership, insolvency, liquidation, bankruptcy, reorganization, arrangement, adjustment, composition or other judicial proceeding relative to a Borrower or any other of such other obligor of the Note or the property of a Borrower or of such other obligor or their creditors, the Bank or the Agent (irrespective of whether the principal of the Note shall then be due and payable as therein expressed or by declaration or otherwise and irrespective of whether the holder of the Note shall have made any demand on a Borrower for the payment of overdue principal, interest or premium) shall be entitled and empowered, by intervention in such proceeding or otherwise to file and prove a claim for the whole amount of principal and interest owing and unpaid in respect of the Note and to file such other papers or documents as may be necessary or advisable in order to have the claims of the holder of the Note allowed in such judicial proceeding.

#### SETOFFS, ETC.

Any Indebtedness from any Bank to Borrowers may be offset and applied toward the payment of any Indebtedness from Borrowers to such Bank, whether or not such Indebtedness, or any part thereof, shall then be due.

The Banks agree between themselves that, with respect to all sums received by the Banks applicable to the payment of principal of or interest on the Notes, equitable adjustment will be made between the Banks so that, in effect, all such sums shall be shared ratably by each of the Banks whether received by voluntary payment, by the exercise of the right of setoff or bankers' lien, by counterclaim or crossclaim or by the enforcement of any or all of the Notes. If a Bank receives any payment on its Notes of a sum or sums in excess of its appropriate portion, then such Bank receiving such excess payment shall purchase for cash, at par plus accrued interest, from the other Bank an interest in its Notes in such amounts as shall result in an appropriate participation by each of the Banks in the aggregate unpaid amount of the Notes then outstanding; provided, however, that if all or any portion of such excess payment is thereafter recovered from such Bank, the purchase shall be rescinded and the purchase price restored to the extent of such recovery, but without interest. Borrowers expressly consent to the foregoing arrangements and agree that each Bank so purchasing a participation may exercise all rights of payment (including, without limitation, all rights of setoff, bankers' lien or counterclaim) with respect to such portion as fully as if such Bank were the direct holder of such portion.

## ARTICLE XI

## EXPENSES

Whether or not the transactions contemplated hereby shall be consummated, Borrowers agree to pay: (a) all reasonable expenses, including fees and disbursements of counsel for the Agent and the Banks, which the Agent or the Banks have incurred in connection with the preparation of the Revolving Loan Documents and all other documents related thereto and (b) all reasonable expenses, including fees and disbursements of respective counsel for the Banks, which the Agent or the Banks may hereafter incur in connection with the Revolving Loan Documents and all other documents related thereto (including any amendment, consent or waiver hereafter requested by Borrowers hereunder or thereunder) and the transactions contemplated hereby or the enforcement of the rights of the Agent or the Banks hereunder or under the Revolving Loan Documents in the event of a default thereunder by Borrowers.

## ARTICLE XII

### AMENDMENTS AND WAIVERS, ETC.

12.1 Amendment. Any term of this Agreement or of the Notes may be amended and the observance of any term of the Revolving Loan Documents may be waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of Borrowers and the Agent; provided, however, that no such amendment or waiver shall, without the express prior written consent of all of the Banks (i) extend the fixed maturity or reduce the principal amount of, or reduce the rate or extend the time of payment of interest on, or reduce the amount or extend the time of payment of any principal of, any Note; (ii) reduce the amount or extend the time of payment of any fee payable hereunder; (iii) change the Commitment of any Bank; (iv) release the security interest in, or assignment with respect to, any Collateral or the Leased Real Property; or (v) amend this Article XII. Any amendment or waiver effected in accordance with this Article XII shall be binding upon each holder of the Note at the time outstanding, each future holder of the Note and Borrowers.

12.2 No Waiver. Any Bank's or the Agent's failure to insist upon the strict performance of any term, condition or other provision of the Revolving Loan Documents or to exercise any right or remedy hereunder or thereunder shall not constitute a waiver by such Bank or the Agent of any such term, condition or other provision or default or Event of Default in connection therewith; and any such waiver shall not affect or alter the Revolving Loan Documents, and each and every term, condition and other provision of the Revolving Loan Documents, in such event, shall continue in full force and effect and shall be operative with respect to any other existing or subsequent default or Event of Default in connection therewith.

### ARTICLE XIII

## PARTICIPATIONS

At any time and from time to time, each Bank may freely sell participations in its Note or assign its interests under the Revolving Loan Documents; provided, however, that no assignee, participant or transferee of NationsBank's rights shall be entitled to receive any greater payments under Sections 2.10 and 2.11 than NationsBank would have been entitled to receive with respect to the rights assigned, participated or otherwise transferred. Borrowers shall execute and deliver such documents as may be required to effect any such participation or assignment and shall pay to the Agent an agent fee equal to \$50,000 per year or any portion thereof, payable in advance on the date such participation or assignment is made and on each anniversary thereof.

#### ARTICLE XIV

#### INDEMNIFICATION

14.1 Indemnification Under Agreement. Borrowers shall indemnify each Bank and the Agent, and their respective officers, directors, employees, Affiliates and agents (each, an "Indemnified Party") for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses (including reasonable attorneys' fees, including allocated cost of in-house counsel) or disbursements of any kind and nature whatsoever which may be imposed on, incurred by or asserted against such Indemnified Party in any way relating to or arising out of any Revolving Loan Document or any other document contemplated by or referred to therein or the transactions contemplated by or referred to therein, the enforcement of any of the terms of any Revolving Loan Document or any such other documents; provided, however, that Borrowers shall not be liable for any of the foregoing to the extent they arise from (i) the gross negligence or willful misconduct of such Indemnified Party, (ii) any actions or suits which are brought by such Bank against the Agent or (iii) any actions or suits which are brought by the Agent against such Bank (the "Excluded Claims").

14.2 Indemnification Relating to Legal Actions. Should any of the Agent's or any Bank's officers or employees be involved in any legal action or proceeding in connection with the transactions contemplated hereby (other than relating to an Excluded Claim), Borrowers shall compensate the Agent or such Bank in a reasonable amount to be mutually agreed upon per officer or employee per day for each day or portion thereof that such employee is involved in preparation and testimony pertaining to any such legal action or proceeding. The Agent or such Bank, as the case may be, shall give Borrowers prompt written notice of any claim setting forth a description of those elements of the claim of which the Agent or such Bank has knowledge. Borrowers shall have the right at any time during which a claim is pending to select counsel to defend and settle any claims for which Borrowers are indemnitors hereunder so long as in any such event Borrowers shall have stated in a writing delivered to such Bank or the Agent that, as between Borrowers and such Bank or the Agent, Borrowers are responsible to such Bank or the Agent with respect to such claim to the extent and subject to the limitations set forth herein. In any other case, such Bank or the Agent shall have the right to select counsel and control the defense of any claims to the extent provided for herein; provided, however, that neither such Bank nor the Agent shall settle any claims as to which it is controlling the defense without Borrowers' consent, which consent shall not be unreasonably withheld. With respect to any claim for which Borrowers are entitled to select counsel, such Bank or the Agent shall have the right, at its expense, to participate in the defense of such claim.

14.3 Contribution. If for any reason the foregoing indemnity is unenforceable by any Indemnified Party or insufficient to hold it or them free and harmless as contemplated by the preceding clauses, then Borrowers shall contribute to the amount paid or payable by such Indemnified Party as a result of any claim (other than an Excluded Claim) in such proportion as is appropriate to reflect not only the relative benefits received by Borrowers on the one hand and such Indemnified Party on the other hand, but also the relative fault of Borrowers and such Indemnified Party, as well as any other relevant equitable considerations.

14.4 No Interference with Other Agreements. The obligations of Borrowers under this Article XIV shall survive payment and performance of all other Obligations to the Banks and are in addition to, and shall not otherwise limit, any warranties or similar obligations of Borrowers or any Affiliate of Borrowers in any other agreement or instrument.

## ARTICLE XV

## THE AGENT

15.1 Appointment, Powers and Immunities. The Banks hereby irrevocably appoint and authorize the Agent to act as their agent hereunder and under each of the other Revolving Loan Documents with such powers as are specifically delegated to the Agent by the terms of the Revolving Loan Documents, together with such other powers as are reasonably incidental thereto. The Agent (which term shall include reference to its Affiliates and its own and its Affiliates' officers, directors, employees and agents): (a) shall have no duties or responsibilities except those expressly set forth in the Revolving Loan

Documents; (b) shall not be responsible to the Banks for any recitals, statements, representations or warranties contained in any Revolving Loan Document, or in any certificate or other document referred to or provided for in, or received by any of them under, any Revolving Loan Document, or for the value, validity, effectiveness, genuineness, enforceability, perfection or sufficiency of any Revolving Loan Document or any other document referred to or provided for or for any failure by Borrowers or any other Person to perform any of its obligations thereunder; (c) shall have the right, at its option, in an Event of Default of Borrowers hereunder, to purchase the Note of any Bank at par, plus accrued interest; (d) shall not be required to initiate or conduct any litigation or collection proceedings hereunder; and (e) shall not be responsible for any action taken or omitted to be taken by it hereunder or under any other document or instrument referred to or provided for herein or in connection herewith, except for its own gross negligence or willful misconduct. The Agent may employ agents and attorneys-in-fact and shall not be responsible for the negligence or misconduct of any such agents or attorneys-in-fact selected by it with reasonable care. Subject to the foregoing, the Agent shall, on behalf of the Banks: (i) execute any and all of the Revolving Loan Documents on behalf of the Banks, except where the Banks are parties thereto, (ii) hold and apply the Pledged Stock, and the proceeds thereof, in accordance with the terms of the Pledge Agreements and this Agreement; (iii) hold and apply any and all Collateral, and the proceeds thereof, at any time received by it, in accordance with the provisions of the Security Documents and this Agreement; (iv) exercise any and all rights, powers and remedies of the Banks under any Revolving Loan Document, including the giving of any consent or waiver or the entering into of any amendment, subject to the provisions of Section 12.1; (v) execute, deliver, file and possess instruments on behalf of any or all of the Banks; and (vi) in the event of acceleration of Borrowers' indebtedness hereunder, use its best efforts to sell or otherwise liquidate or dispose of the Collateral referred to herein, in the Security Documents and otherwise exercise the rights of the Banks thereunder.

15.2 Reliance. The Agent shall be entitled to rely upon any certifications, notices or other communications (including any communications by telephone, telex, telegram or cable) reasonably believed by it to be genuine and correct and to have been signed or sent by or on behalf of the proper Person or Persons, and upon advice and statements of legal counsel, independent accountants and other experts selected by such Agent. As to any matters not expressly provided for by this Agreement, the Agent shall in all cases be fully protected in acting, or in refraining from acting, hereunder in accordance with instructions signed by the Banks, and such instructions of the Banks and any action taken or failure to act pursuant thereto, shall be binding on the Banks.

15.3 Defaults. The Agent shall not be deemed to have knowledge of the occurrence of an Event of Default (other than the nonpayment of principal of or interest on the Notes) unless it has received notice from a Bank or Borrowers specifying such Event of Default. In the event that the Agent receives such a notice of the occurrence of an Event of Default, it shall give prompt notice thereof to the Banks (and shall give each Bank prompt notice of each such nonpayment). The Agent shall (subject to Section 15.7) take such action, or refrain from taking such action, with respect to such Event of Default as it shall deem advisable in the best interest of the Banks.

15.4 Rights as a Bank. With respect to the Commitment and the Loans made by it, the Agent, in its capacity as a Bank hereunder, shall have the same rights and powers hereunder as the other Banks and may exercise the same as though it were not acting as the Agent, and the term "Bank" or "Banks" shall, unless the context otherwise indicates, include the Agent in its individual capacity. The Agent and its Affiliates may (without having to account therefor to the other Banks) accept deposits from, lend money to and generally engage in any kind of banking, trust or other business with Borrowers and any of its Affiliates as if it were not acting as the Agent, and the Agent may accept fees and other consideration from the Borrowers for services as the Agent or otherwise without having to account for the same to the other Banks.

15.5 Indemnification. The Banks agree to indemnify the Agent ratably in accordance with the aggregate principal amount of the Notes held by the Banks (or, if no such principal or interest is at the time outstanding, ratably in accordance with their respective Commitments), for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind and nature whatsoever (including, without limitation, the costs and expenses which Borrowers are obligated to pay but excluding, unless an Event of Default has occurred and is continuing, normal administrative costs and expenses incident to the performance of its agency duties hereunder) which may be imposed on, incurred by or asserted against the Agent in any way relating to or arising out of any Revolving Loan Document or any other document contemplated by or referred to herein or therein, or the enforcement of any of the terms of any Revolving Loan Document or of any such other documents; provided, however, that no Bank shall be liable for any of the foregoing to the extent they arise from the gross negligence or willful misconduct of the party to be indemnified.

15.6 Non-Reliance on Agent and Other Banks. Each Bank agrees that it has, independently and without reliance on the Agent or the other Banks, and based on such documents and information as it has deemed appropriate, made its own credit

analysis of Borrowers and its own decision to enter into this Agreement and that it will, independently and without reliance upon the other Banks, and based on such documents and information as it shall deem appropriate at the time, continue to make its own analysis and decisions in taking or not taking action under this Agreement. The Agent shall not be required to keep itself informed as to the performance or observance by Borrowers of this Agreement or any other document referred to or provided for herein or to inspect the properties or books of Borrowers. Except for notices, reports and other documents and information expressly required to be furnished to the Banks by the Agent hereunder, the Agent shall not have any duty or responsibility to provide any Bank with any credit or other information concerning the affairs, financial condition or business of Borrowers which may come into the possession of the Agent or any of its Affiliates. Notwithstanding the foregoing, the Agent will use its best efforts to provide to the Banks any and all information reasonably requested by them and reasonably available to the Agent promptly upon such request.

15.7 Failure to Act. Except for actions expressly required of the Agent hereunder, the Agent shall in all cases be fully justified in failing or refusing to act hereunder unless it shall be indemnified to its satisfaction by the Banks against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action.

15.8 Resignation of Agent. Subject to the appointment and acceptance of a successor Agent as provided below, the Agent may resign at any time by giving notice thereof to the Banks and Borrowers. Upon any such resignation, the Banks shall have the right to appoint a successor Agent without the consent of Borrowers. Such successor Agent so appointed by the Banks shall be a bank which has a combined capital and surplus of at least \$75,000,000. Upon the acceptance of any appointment as the Agent hereunder by a successor Agent, such successor Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Agent, and the retiring Agent shall be discharged from its duties and obligations hereunder. After the retiring Agent's resignation hereunder as Agent, the provisions of this Article XV shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as the Agent.

15.9 Cooperation of Banks. Each Bank shall (a) promptly notify the other Banks and the Agent of any Event of Default known to such Bank under this Agreement and not reasonably believed to have been previously disclosed to the other Banks and the Agent, (b) provide the other Banks and the Agent with such information and documentation as the other Banks or the Agent shall reasonably request in the performance of their respective duties hereunder, including all information relative to the outstanding balance of principal, interest and other sums owed to any such Bank by Borrowers; and (c) cooperate with the Agent with respect to any and all collections and/or foreclosure procedures at any time commenced against Borrowers or otherwise in respect of the Collateral by the Agent on behalf of the Banks.

15.10 Amendment of Article XV. Borrowers hereby agree that the foregoing provisions of this Article XV constitute an agreement among the Agent and the Banks (and the Agent and the Banks acknowledge that Borrowers are not party to or bound by such foregoing provisions and will not be bound by any amendment of such foregoing provisions) and that any and all of the provisions of this Article XV may be amended at any time by the Banks and the Agent without the consent or approval of, or notice to, Borrowers (other than the requirement of notice to Borrowers of the resignation of the Agent).

# ARTICLE XVI

## MISCELLANEOUS

16.1 Notices, etc. All notices and other communications required or permitted to be given hereunder shall be in writing and shall be delivered personally, telegraphed, telexed, sent by facsimile transmission or sent by certified, registered, or express mail, postage prepaid, to the parties at the addresses set forth below (or to such other addresses as the parties may specify by due notice to the others). Notices or other communications given by certified, registered, or express mail shall be deemed given three (3) Banking Days after the date of mailing. Notices or other communications sent in any other manner shall be deemed given only when actually received.

(a) If to Borrowers, to:

Telos Corporation (on behalf of each Borrower) 19886 Ashburn Road Ashburn, Virginia 20147 Attention: President and Chief Financial Officer Telecopy: (703) 724-3855

(b) If to NationsBank, to:

NationsBank, N.A. 8300 Greensboro Drive, Suite 550 McLean, Virginia 22102-3604 Attention: Catherine G. Beddow Vice President Telecopy: (703) 761-8344

with a copy to:

Hogan & Hartson L.L.P. Columbia Square 555 Thirteenth Street, N.W. Washington, D.C. 20004-1109 Attention: Claudette M. Christian, Esq. Telecopy: (202) 637-5910

(c) If to the Agent to:

NationsBank, N.A. 8300 Greensboro Drive, Suite 550 McLean, Virginia 22102-3604 Attention: Catherine G. Beddow Vice President Telecopy: (703) 761-8344

with a copy to:

Hogan & Hartson L.L.P. Columbia Square 555 Thirteenth Street, N.W. Washington, D.C. 20004-1109 Attention: Claudette M. Christian, Esq. Telecopy: (202) 637-5910

(d) If to any other Bank, as set forth in a written notice to each other party hereto or to such other address or addresses as the party to whom such notice is directed may have designated in writing to the other parties hereto.

16.2 Calculations, etc. Subject to the provisions of this Agreement, calculations hereunder shall be made and financial data required hereby shall be prepared, both as to classification of items and as to amounts, in accordance with GAAP, which principles and practices shall be consistently applied and in conformity with those used in the preparation of the consolidated financial statements referred to in Section 6.1.

16.3 Governmental Approval. Borrowers agree to take any related action which any Bank may reasonably request in order to obtain and enjoy the full rights and benefits granted to such Bank by the Revolving Loan Documents.

16.4 Survival of Agreements, etc. The Revolving Loan Documents and the rights and obligations specified therein shall be binding upon and shall inure to the benefit of the parties thereto and their respective successors and assigns, and any subsequent holder or holders of the Note, and the term "Bank" shall include any such holder whenever the context permits, subject to the provisions of Section 13 hereof. All representations and warranties made herein and in the Original Credit Agreement shall survive the execution and delivery of this Agreement and the making of the Loans hereunder until such time as all amounts owing under this Agreement have been paid in full.

16.5 Counterparts, etc. To facilitate execution, this Agreement may be executed in as many counterparts as may be required; and it shall not be necessary that the signatures of, or on behalf of, each party, or that the signatures of all persons required to bind any party, appear on each counterpart; but it shall be sufficient that the signatures of, or on behalf of, each party; or that the signatures of the persons required to bind any party, appear on one or more of the counterparts. Delivery of an executed counterpart of a signature page to this Agreement by facsimile shall be effective as delivery of a manually executed signature page hereto. All counterparts shall collectively constitute a single agreement. It shall not be necessary in making proof of this Agreement to produce or account for more than a number of the parties hereto.

16.6 Amendment. The Revolving Loan Documents may not be changed orally but only by an agreement in writing signed by the party against whom any waiver, change, modification or discharge is sought.

16.7 WAIVER OF JURY TRIAL AND SETOFF; CONSENT TO JURISDICTION. BORROWERS HEREBY WAIVE, TO THE EXTENT PERMITTED BY APPLICABLE LAW, TRIAL BY JURY IN ANY LITIGATION IN ANY COURT WITH RESPECT TO, IN CONNECTION WITH, OR ARISING OUT OF THIS AGREEMENT, ANY OTHER REVOLVING LOAN DOCUMENT, THE COLLATERAL OR ANY INSTRUMENT OR DOCUMENT DELIVERED PURSUANT TO THIS AGREEMENT, OR THE VALIDITY, PROTECTION, INTERPRETATION, COLLECTION OR ENFORCEMENT THEREOF, OR ANY OTHER CLAIM OR DISPUTE HOWSOEVER ARISING, BETWEEN BORROWERS, ON THE ONE HAND, AND THE AGENT OR ANY ONE OR MORE OF THE BANKS, ON THE OTHER HAND; AND BORROWERS HEREBY WAIVE, TO THE EXTENT PERMITTED BY APPLICABLE LAW, THE RIGHT TO INTERPOSE ANY SETOFF OR COUNTERCLAIM OR CROSS-CLAIM IN CONNECTION WITH ANY SUCH LITIGATION, IRRESPECTIVE OF THE NATURE OF SUCH SETOFF, COUNTERCLAIM OR CROSS-CLAIM EXCEPT TO THE EXTENT THAT THE FAILURE SO TO ASSERT ANY SUCH SETOFF, COUNTERCLAIM OR CROSS-CLAIM WOULD PERMANENTLY PRECLUDE THE PROSECUTION OF OR RECOVERY UPON SAME. BORROWERS HEREBY IRREVOCABLY CONSENT TO THE NONEXCLUSIVE JURISDICTION OF THE COURTS OF THE COMMONWEALTH OF VIRGINIA AND, TO THE EXTENT PERMITTED BY APPLICABLE LAW, OF ANY FEDERAL COURT LOCATED IN THE COMMONWEALTH OF VIRGINIA IN CONNECTION WITH ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO ANY ONE OR MORE OF THIS AGREEMENT, ANY OTHER REVOLVING LOAN DOCUMENT, OR ANY DOCUMENT OR INSTRUMENT DELIVERED PURSUANT TO THIS AGREEMENT.

16.8 Governing Law. The Revolving Loan Documents, including the Notes and the validity thereof and the rights and obligations of the parties thereunder, shall be construed in accordance with the laws of the Commonwealth of Virginia (but not including the choice of law rules thereof).

16.9 Time of Essence. The parties acknowledge that time is of the essence in effecting the transactions contemplated hereby.

16.10 Replacement Notes. If any Note shall become mutilated, destroyed, lost or stolen, Borrowers shall, upon the written request of the holder of such Note, execute and deliver in the replacement thereof a new Note in the same form, in the same original principal amount and dated the same date as the Note so mutilated, destroyed, lost or stolen. If such Note has been destroyed, lost or stolen, the holder of such Note shall furnish to Borrowers such security or indemnity as may be reasonably required by Borrowers to save Borrowers harmless and evidence satisfactory to Borrowers of the destruction, loss or theft of such Note and of the ownership thereof; provided, however, that, if the holder of such Note is NationsBank, the written undertaking of NationsBank delivered to Borrowers shall be sufficient security and indemnity.

16.11 Lien. Borrowers hereby grant to each Bank and the Agent a continuing Lien for all obligations under the Revolving Loan Documents and any interest therein upon any and all monies, securities or other property of Borrowers and the proceeds thereof, now or hereafter held or received by, or in transit to, such Bank or the Agent from or for Borrowers, whether for safekeeping, custody, pledge, transmission, collection or otherwise, and also upon any and all deposits (general or special) and credits of Borrowers with, and any and all claims of Borrowers against, such Bank or the Agent at any time existing. Upon the occurrence of an Event of Default, each Bank and the Agent are hereby authorized at any time and from time to time, without notice to Borrowers, to set off, appropriate and apply any or all items hereinabove referred to against all indebtedness of Borrowers to such Bank or the Agent.

16.12 Information. The Agent and each Bank are hereby authorized to deliver any information, reports, communications, notices and financial statements furnished to any of them by or on behalf of Borrowers or any Affiliate of Borrowers or obtained by any of them from Borrowers under this Agreement, to (i) any holder of Subordinated Debt or regulatory body or the Agent or such Bank (to the extent requested by such regulatory body or agency), (ii) any prospective participant in or prospective assignee of the Loans or any portion thereof and which is a financial institution or (iii) any other Person which shall, or shall have any right or obligation to, succeed to all or any part of the interest of the Agent or such Bank in the Revolving Loan Documents, or any security provided for or otherwise securing the obligations of Borrower hereunder.

16.13 Revision of Schedules. From and after the date hereof, Borrowers shall have the limited right to update the Schedules attached to this Agreement provided that any additional information reflected in such update is not material in any respect to Borrowers or to the transactions contemplated by the Revolving Loan Documents.

16.14 Joint and Several Liability. To the extent that more than one person or entity is included in the term "Borrowers," then the liability of all such persons shall be joint and several. In addition, all references to the "Borrowers" shall be interpreted as references to each such Person individually and to all such Persons collectively. IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first above written.

TELOS CORPORATION, formerly known as C3, INC.

By: /s/ William L.P. Brownley

Title: Vice President & General Counsel

TELOS CORPORATION

By: /s/ Lorenzo Tellez

Title: Chief Financial Officer &Treasurer

NATIONSBANK, N.A.

By: /s/ Catherine G. Beddow

Title: Vice President

NATIONSBANK, N.A., as the Agent

By: /s/ Catherine G. Beddow

Title: Vice President

This schedule contains summary financial information extracted from the consolidated balance sheets and statements of income for Telos Corporation and is qualified in its entirety by reference to such financial statements.

```
6-M0S
             DEC-31-1997
                  JUN-30-1997
                             757,000
                              0
               50,079,000
1,047,000
15,673,000
69,548,000
                          38,116,000
                21,580,000
105,748,000
         36,682,000
                         50,068,000
         43,577,000
                               0
                             78,000
                     (36,914,000)
105,748,000
                         53,134,000
              112,434,000
                         45,512,000
                  95,159,000
0
                         0
              3,643,000
                  (173,000)
                              0
              (173,000)
                           0
                           0
                                  0
                      (173,000)
                             0
                             0
```